

Temi di Discussione

(Working Papers)

Mapping local productivity advantages in Italy: industrial districts, cities or both?

by Valter Di Giacinto, Matteo Gomellini, Giacinto Micucci and Marcello Pagnini







Temi di discussione

(Working papers)

Mapping local productivity advantages in Italy: industrial districts, cities or both?

by Valter Di Giacinto, Matteo Gomellini, Giacinto Micucci and Marcello Pagnini

Number 850 - January 2012

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

This paper is part of a research program on "The transformation of local production systems" whose results were presented and discussed at a conference held in Bologna on January 31 and February 1, 2012, organized by the Bank of Italy and the Department of Economics, University of Bologna.

Editorial Board: Silvia Magri, Massimo Sbracia, Luisa Carpinelli, Emanuela Ciapanna, Francesco D'Amuri, Alessandro Notarpietro, Pietro Rizza, Concetta Rondinelli, Tiziano Ropele, Andrea Silvestrini, Giordano Zevi. Editorial Assistants: Roberto Marano, Nicoletta Olivanti.

MAPPING LOCAL PRODUCTIVITY ADVANTAGES IN ITALY: INDUSTRIAL DISTRICTS, CITIES OR BOTH?

by Valter Di Giacinto*, Matteo Gomellini*, Giacinto Micucci* and Marcello Pagnini*

Abstract

We compare the magnitude of local productivity advantages associated with two different spatial concentration patterns in Italy – urban areas and industrial districts. The former have high population density and host a wide range of economic activities, while the latter are marked by a high concentration of small firms producing relatively homogenous goods. Using data from a large sample of Italian manufacturing firms observed over the 1995-2006 period, we detect local productivity advantages for both urban areas and industrial districts. However, firms located in urban areas reap a larger productivity premium than those operating within districts. The advantages of industrial districts have declined over time; those of urban areas have remained stable. Differences in the composition of firm employees between white- and blue-collars explain a small fraction of the urban productivity premium. The quantile regressions show how more productive firms gain larger benefits by locating in urban areas to prosper in a world characterized by advancing globalization and the growing use of ICT.

JEL Classification: C52, D24, R12.

Keywords: urban areas, industrial districts, agglomeration economies, productivity, whiteand blue-collars, Italian economy.

Contents

1. Introduction	5
2. Industrial districts and urban areas as sources of local productivity advantages	7
3. Industrial districts and urban areas in Italy: definitions and structural differences	10
4. Data	12
5. TFP estimation	14
6. Estimation of TFP differentials	16
7. Discussion of the main results and further extensions	20
7.1 The evolution of local productivity advantages in urban areas and industrial	
districts	20
7.2 The role of human capital	21
7.3 Quantile analysis	23
8. Conclusion	26
Tables	29
Appendix 1 - Imputing employee data	41
Appendix 2 - The relation between TFP and firm size	41
Appendix 3 - Additional robustness checks	42
References	51

^{*} Bank of Italy, Economic Research Unit, L'Aquila, Roma, Ancona and Bologna Branch. E-mail: <u>valter.digiacinto@bancaditalia.it</u>, <u>matteo.gomellini@bancaditalia.it</u>, <u>giacinto.micucci@bancaditalia.it</u>, <u>marcello.pagnini@bancaditalia.it</u>.

1. Introduction¹

The forces driving spatial agglomeration manifest themselves in different ways even within the same country or sector of economic activity. Urban areas typically have high population density, a wide range of economic activities, including a highly diversified service sector, and extensive local amenities coupled with high congestion costs. Industrial clusters or districts, instead, are usually located outside of urban areas, show a high concentration of small firms producing relatively homogenous goods and, albeit in a different way, may also be affected by congestion (for surveys of the empirical literature on agglomeration economies, see Rosenthal and Strange, 2004, and Melo, Graham and Noland, 2009).

In this paper we address several questions concerning these two spatial concentration patterns with regard to the Italian economy: i) Are plants located in urban areas and industrial districts more productive than firms located elsewhere? ii) Are the local productive advantages in the two spatially concentrated areas comparable in magnitude? iii) How have these advantages changed in recent years?

Answering the first question may shed light on the mechanisms that generate agglomeration economies, a long-debated issue in the literature. The second question is relatively new and especially relevant in the context of the Italian economy. Finally, the third question aims at documenting how the comparative advantages of urban areas and industrial districts have evolved in the new environment shaped by increasing competition from newly industrialized countries and the advent of information and communication technologies (Glaeser and Ponzetto, 2010).

The empirical literature on agglomeration economies has usually addressed similar questions by regressing average productivity across areas on a series of explanatory variables including local market size, usually proxied by population or population density, the sectoral diversification of the local economy, its relative specialization in a specific sector and the share of small firms. In this context, positive partial correlation between productivity and market size or diversity is usually interpreted as evidence that urbanization is responsible for agglomeration economies, while a positive coefficient for the specialization indicator or the share of small

¹ The authors wish to thank the following for helpful comments: Roberto Camba, Luigi Cannari, Davide Castellani, Gilles Duranton, Andrea Filippone, Giovanni Iuzzolino, two anonymous referees, and participants in seminars held at the Bank of Italy, University of Toronto, University of Jena, University of Rimini, University of Parma, University of Milan (Italian Trade Study Group) and University of Barcelona (ERSA annual meeting). The views expressed in this paper are our own and do not necessarily reflect those of the Bank of Italy.

firms is taken as a signal that spatial clustering in the industrial districts is the main driver of the local productivity advantages.

In this paper we take a slightly different route by mapping the Italian territory into three non-overlapping areas: a) urban areas, defined as locations with a population above a certain threshold; b) industrial districts, identified through a complex algorithm that will be defined later in the paper; and c) other locations. We then measure average local productivity differentials by regressing firm-level indicators of productive efficiency on UA and ID dummies plus a set of controls.²

Apart from permitting a straightforward comparison of the magnitudes of productivity gains associated with industrial districts and urban areas, the advantages of this empirical strategy are manifold. Good proxies for the positive externalities associated with urban areas are usually difficult to devise and are in any case related to the fact that the local population has to be above a certain threshold for these agglomeration forces to produce their effects (this consideration equally applies to negative externalities, namely congestion effects). The identification of industrial districts is also quite complex. In Italy, an official definition of industrial district is produced by the National Statistical Institute (Istat) as the outcome of a multi-step algorithm. Since mimicking that algorithm in a regression analysis using a set of continuous variables would be both demanding and inefficient, we chose to summarize the complex structural characteristics of Italian industrial districts by means of a dummy variable that singles out the local labor markets classified as districts in Istat's taxonomy.

For our inquiry we use a panel of 29,000 Italian manufacturing firms observed over the period 1995-2006. Our main findings are the following: The two different spatial concentration patterns associated with urban areas and industrial districts are both able to generate local productivity advantages; however, the advantages are greater in urban areas. Moreover, comparative advantages in cities remained stable over the period 1995-2006, while those in the industrial districts declined. In addition, productivity advantages persist in urban areas even controlling for differences in workforce composition. Blue-collar workers are more productive in industrial districts, while white-collar workers, whose role in upgrading production is increasingly important, show higher productivity in urban areas. Finally, a quantile regression is used to show that in industrial districts there is a stronger positive impact on the lower tail of the TFP distribution, while urban areas benefit more firms belonging to the upper tail. Several

² For a survey of recent empirical work on productivity differentials across firms, see Syverson (2010).

shocks – the introduction of the euro, the rapid diffusion of ICT and growing globalization – affected the Italian economy at the turn of the century. Our results suggest that urban areas responded more effectively than industrial districts to those developments.

The rest of the paper is organized as follows. Section 2 presents a brief review of literature investigating the importance of agglomeration effects for firms' productivity. Sections 3 and 4 discuss, respectively, the territorial level of analysis and the data. Section 5 reports the TFP estimation. Section 6 analyses the impact of spatial concentration on firms' TFP. In Section 7 several extensions to the baseline results are proposed including the effects of human capital heterogeneity on firms' productivity. Section 8 concludes.

2. Industrial districts and urban areas as sources of local productivity advantages

Spatial concentration may generate local productivity advantages through different mechanisms. Agglomeration economies in the form of technical or knowledge spillovers, labor market pooling and proximity to local buyers or sellers may increase the productivity of firms located in densely populated areas. In addition, recent studies point to a mechanism based on selection: large markets attract more entrants, thereby fostering competition and inducing less efficient firms to exit the local market.³ Finally, other contributions stress the sorting of firms or workers.⁴ Ex ante heterogeneous firms may differ in their ability to exploit local productivity advantages; for instance, less efficient firms may avoid the harshness of competition by locating in less dense areas, while large and more productive firms may be better able to exploit the benefits of different kinds of agglomeration economies.

Urban areas and industrial districts are examples of geographical concentration with different characteristics. The question we want to address is whether and to what extent these differences will be reflected in the strength of local productivity advantages generated by the two environments.

³ For this class of models see Melitz (2003) and Melitz and Ottaviano (2008). Syverson (2004a) analyses the effects of local market size on productivity and firm selection in the special case of the concrete industry, where transport costs are significant. Syverson (2004b) and Del Gatto, Ottaviano and Pagnini (2008) investigate how selection effects vary across different industries in response to a set of their characteristics (elasticity of demand, openness to trade). Their implicit assumption is that markets in many branches of manufacturing are integrated through domestic trade.

⁴ The fast-growing literature on sorting focuses mainly on workers; see Combes, Duranton and Gobillon (2008) for France, and Matano and Naticchioni (2011) for Italy. For the theoretical literature on firm sorting, see also Baldwin and Okubo (2006) and Okubo, Picard and Thisse (2010). Nocke (2006) pursues a similar line of research, starting out, however, from the tenets of oligopoly theory.

In the view of some scholars, Italy's industrial boom after the Second World War was triggered by the growth of industrial districts, areas with a high concentration of small firms cooperating along the production chain of a unique final good.⁵ Industrial districts usually have a strong specialization in manufacturing, and the district "community" may also include local institutions, political parties, associations and local banks.

Urban areas, with their high population density, attract a diversified set of activities, including transport, recreational and other services. This also stimulates the production of local amenities (cultural activities) and disamenities (pollution and so on).

Given these characteristics, it is likely that both urban areas and industrial districts will be able to generate some agglomeration economies. An important question is whether these will be produced by the interactions of firms and workers within the same industry (Marshall externalities) or, alternatively, belonging to different industries (Jacob externalities). Quite clearly, Marshall externalities are typical of industrial districts, while Jacob externalities are more likely to arise in urban areas.

To avoid the paradoxical outcome of an economy concentrated in just one type of region, these local productivity advantages have to be traded off against other factors whose nature depends on the kind of the production process and that may induce firms to locate outside industrial districts and urban areas. Congestion costs, for instance, can lead to resource mismanagement within a firm, thereby lowering production efficiency in cities.⁶ Although industrial districts can save on congestion costs thanks to their specialization in a specific industry, they can still be exposed to the problems caused by the crowding of firms and workers in a relatively small area. Moreover, their productivity advantages can diminish when indivisibilities are important. In those circumstances, the network externalities generated within large and hierarchical organizations. Finally, these sources of local comparative advantages may change over time because of the evolution of technology or changes in the domestic or international competitive environment (liberalizations and so on).

⁵ Becattini (1990) provides a conceptualization of the industrial district, defining it as a socio-territorial entity characterized by the active presence of both a community of people and a population of firms in one naturally and historically bounded area. Thus, an economic definition of the industrial district that aims to be comprehensive will have to include both the network of links between firms and the above mentioned social conditions. For a recent survey and empirical analysis on Italian districts, see Iuzzolino and Micucci (2011).

⁶ Moreover, they can push up local land prices, so that firms whose production is land-intensive will be induced to locate outside urban areas.

As for the selection effects and sorting, it is difficult to say a priori whether they will be stronger in industrial districts and urban areas. We will discuss them in Section 7.

The empirical literature on the sources of local productivity advantages analyses the effects of urban areas mainly through the size of the local market. A positive correlation between market size and productivity is usually interpreted as evidence that cities promote production efficiency. Doubling city size would increase productivity by between 3 and 8 per cent, depending on the study and the country considered.⁷ As far as we know, no paper has estimated that elasticity for Italy. The contribution that comes closest is that of Cingano and Schivardi (2005), who show that moving from the first to the third quartile of city-size distribution would increase the annual growth of total factor productivity (TFP) by 0.6 per cent for a sample of Italian manufacturing firms.

Whereas the contributions referring to other countries emphasize urban effects, the empirical literature in Italy has focused mainly on the productivity advantages associated with industrial districts.⁸ In particular, Signorini (1994; see Table 1), using data for the provinces of Prato and Biella, finds that firms in districts have higher per capita value added. Fabiani et al. (2000) extend the analysis to all of Italy, showing that between 1982 and 1995 district firms outperformed non-district firms. In 1995, district firms' return on investment was higher by 2 percentage points and their return on equity by 4.1 points. Value added per worker was 1.3 per cent higher on average in district firms. Furthermore, district firms were less inefficient than non-district firms in 8 out of the 13 sectors considered.⁹

Cainelli and De Liso (2005) estimate the effects of clustering of the firms in districts on productivity. Disentangling process and product innovation, they trace most of the productivity advantages for district firms to the latter. They find that the district effect, measured as the differential in value added growth rates, ranges between 2.0 and 2.6 percentage points.

Cingano and Schivardi (2005) offer indirect evidence of a positive district effect by showing that increasing local sectoral specialization (a characteristic associated with industrial

⁷ Rosenthal and Strange (2004). See also Melo, Graham and Noland (2009) for a survey of this literature and for a meta-analysis of the relation between productivity and city size.

⁸ For a short review of the papers assessing industrial district advantages, see the list in Table 1.

⁹ The authors use a stochastic frontier approach to measure inefficiency. They define technical inefficiency as "the failure to produce the maximum possible output for any chosen combinations of inputs", including "the inefficiency arising from the managerial and organisational structure and the socio-economic environment in which firms operate". Fabiani et al., (2000), p. 58.

districts) would raise local TFP growth by between 0.2 and 0.5 per cent, depending on the specification adopted (according with their results TFP growth is enhanced by specialization and city size but not by urban diversity). Despite this nearly unanimous agreement, the most recent studies have shown that districts' localization advantages are diminishing (as globalization reduces district externalities). Examining the internal features of industrial districts, major recent structural changes can affect their evolution in the future.¹⁰ Foresti, Guelpa and Trenti (2009) use balance sheet indicators for a wide sample of manufacturing firms (unbalanced panel) over the period 1991-2006. Controlling for different characteristics (e.g. sectoral specialization, size) they find signs of a fading of the district effect from the late 1990s onwards.

3. Industrial districts and urban areas in Italy: definitions and structural differences

To assess the existence of local productivity advantages, it is necessary first to map industrial districts and urban areas. In Italy, industrial districts are officially defined by Istat using a multistep algorithm. Although this method is not without flaws, it rapidly became a sort of benchmark for assessing the so-called industrial district premium, i.e. the productivity gain associated with being located in an industrial district. Here we will then describe the methodology used to define these areas.

The starting point is the data on daily commuting flows from place of residence to place of work available for Italy's 8,100 municipalities. Contiguous municipalities are then aggregated into larger areas called Local Labor Market Areas (LLMAs) using a procedure that maximizes labor mobility within these areas and minimizes it between them. This procedure was used to map the Italian territory into 784 LLMAs in 1991 (686 in 2001).¹¹ Notice that LLMA's represent an ideal partition to analyze many agglomeration effects since most of them are conveyed though the interactions taking place within the local labor market. However, this zoning system can be sometimes problematic as far as the definition of the relevant market for manufacturing products is concerned (more on this below).

Industrial districts are defined as LLMAs that satisfy the following conditions:

¹⁰ On the districts' structural evolution, see also Carabelli, Rabellotti and Hirsch (2009).

¹¹ We conduct our empirical analysis on the basis of the 1991 map, which allows us to use a classification that is predetermined with respect to the sample period. This reduces simultaneity problems, due to possible feedback effects from local productivity dynamics on the likelihood that a LLMA is classified as an industrial district. However, using the 2001 map does not basically affect our main results.

a) specialization in the manufacturing sector, i.e. $l_a = \frac{(x_{am}/x_a)}{(x_{\bullet m}/x_{\bullet \bullet})} > 1$ where x_{am}

denotes the number of employees in area a and in all the local manufacturing industries, x_a denotes total employment (including in services and construction) in the area, and $X_{\bullet m}, X_{\bullet \bullet}$ are the corresponding figures at national level.

- b) $s_a = \frac{\left(x_{am}^{small} / x_{am}\right)}{\left(x_{\bullet m}^{small} / x_{\bullet m}\right)} > 1$ where the upper index "small" indicates the number of employees in small and medium-sized enterprises.
- c) Let $l_{as} = \frac{(x_{as} / x_{am})}{(x_{\bullet s} / x_{\bullet m})}$ denote the location quotient for each specific manufacturing industry *s* and define the "dominant manufacturing industry" *d* as the one for which $l_{ad} > 1$ and the level of employment is maximum among the local specialized industries. For *d*, the following condition must hold: $s_{ad} = (x_{ad}^{small} / x_{ad}) > .5$.
- d) Finally, where there is only one medium-sized enterprise, small enterprises' employment share must exceed half that of the medium-sized firm.

Put simply, according to this definition industrial districts are LLMAs where small and medium-sized enterprises account for a significant share of employment both in manufacturing as a whole and in the sector of specialization. Notice that condition a) nearly automatically rules out the possibility that an urban area can be defined as an industrial district, since the former are usually characterized by an extensive presence of services.

As for the mapping of urbanization in Italy, we use a very simple definition: urban areas are LLMAs with a resident population of more than 500,000. Although Italy was historically known as the "land of a hundred cities", it has not seen the growth of mega-cities like those found in several industrial and emerging countries. Setting a relatively low threshold level to define urban areas is therefore consistent with Italy's relatively low degree of urbanization. Using these categories we obtain three non-overlapping sets of localities (the third set is defined as a complement with respect to the groups of LLMAs included in industrial districts and urban areas; Figure 1). Only Padua had characteristics matching both the definition of industrial district and urban area; we opted to include it in the set of industrial districts. In 1991 the algorithm singled out 199 industrial districts (out of 784 LLMAs); in 2001 the number fell to 156 (out of 686). As the map clearly shows, a prominent spatial feature of clusters in Italy is their location almost exclusively in the North and Centre. As for the spatial distribution of urban areas, they are distributed more evenly throughout the country.





4. Data

The empirical analysis presented in this paper was conducted on a large panel of approximately 29,000 Italian manufacturing firms (not plants), observed over the period 1995-2006, and constructed as follows.

Annual balance sheet figures on value added, consumption of intermediate goods, fixed investment and capital stock were drawn from the Chamber of Commerce-Company Accounts Data Service database (Centrale dei Bilanci / Cerved). Additional firm-level data, including the sector of economic activity (up to the 4-digit SIC sector classification), firm location (municipality where the firm is established) and number of employees were also included as auxiliary information in the database.

The information on the municipalities where firms are located allows us to map them into the 784 LLMAs and hence into the three area types described above. Only one third of the firms in the database report employment data. To overcome this shortcoming, missing employment figures were imputed by means of a statistical procedure using total labor costs as the main auxiliary information (see Appendix 1 for methodological details). Unlike the number of employees, data on total labor costs are available for all the sample firms.

The capital stock at firm level was estimated from the book value of investment using the permanent inventory method and deriving the sector-specific depreciation rates from Italian national accounts data. The capital stock in the initial year was estimated using the deflated book value, adjusted for the average age of capital calculated from cumulated depreciation (see Bond et al., 1997). Nominal value added and consumption of intermediate goods figures were deflated using industry-specific price indexes.

Firms with fewer than 5 employees were removed from the sample, since the data were very noisy for that size class. Our final dataset comprises 392,874 observations, nearly equally distributed over the two sub-periods (1995-2000 and 2001-2006; see Table 2). Due to the exclusion of some outliers (see more on this below), we actually use 344,353 observations in our econometric analysis: this means that we have on average about 28,700 firms per year, a very large sample compared with those used in all previous contributions on this topic.

Slightly more than a half of the observations refer to firms in industrial districts and nearly a quarter to urban areas. Consistently with the characteristics of the entire population (see Istat, 2006) the share of firms located in the South is quite small in both the urban area and the industrial district samples.

On average, firms in urban areas have 77.5 employees, as against 43.9 for the district firms and 54.4 for non-urban, non-district firms. Clearly, our sample is partially biased toward large firms, especially if one recalls the prevalence of small enterprises in Italian manufacturing. This characteristic mainly reflects our decision to drop firms with fewer than 5 employees from the sample; to a lesser extent it is also related to the fact that the Centrale dei Bilanci / Cerved database does not include certain categories of firms, e.g. sole proprietorships. Although we are aware that this requires some caution in the interpretation of our results, we strongly prefer dropping firms below the 5-employee threshold in order to preserve data quality. Average firm size for the entire sample fell from 88 to 67 employees between the two sub-periods while it remained constant in the industrial districts (Table 3).

As for ranking areas by labor productivity, the descriptive statistics show that in the North of Italy firms in industrial districts have higher per capita value added than firms in non-agglomerated areas, but lower than those in urban areas. In the Centre and South, district firms trail both urban-area firms and non-district, non-urban firms (Table 4).

The North-South gap in labor productivity (25-30 percent, in line with other studies¹²) emerges in all three groups of areas (districts, urban areas, other LLMA's) and is largest for industrial districts. The sectoral distribution shows that about 45 per cent of the observations are related to the metal and metal products, mechanical and machinery, textiles and apparel industries.

5. TFP estimation

Our estimation strategy proceeds in several steps. First, production function estimates at firm-level are obtained using different methodologies and total factor productivity (TFP) for each firm is computed as the residual of the estimated production function. Second, firm-level TFP estimates are regressed on a set of independent variables to uncover productivity differentials across the three groups of areas defined in the previous section.

In order to derive individual TFP measures, the following standard Cobb-Douglas production function was considered:

$$Q_{i\in(r,s)t} = \Phi_{it} L_{it}^{\alpha_s} K_{it}^{\beta_s} \quad (1)$$

where *L* and *K* denote labor and capital inputs used to produce the amount of output Q in the year *t* by firm *i* belonging to sector *s* and located in LLMA r^{13} ; α_s and β_s are the production function coefficients, which are allowed to vary across sectors.

¹² The North-South divide is an issue of paramount importance for the Italian economy (see for example: Cannari, Magnani and Pellegrini, 2009). Investigating it in relation to the geographical distribution of agglomeration economies could be rewarding but is beyond the scope of the present paper.

¹³ To avoid cluttering notation, in the following we drop the reference to the LLMA and the sector when indexing variables referring to the individual firm.

After log transformation the following estimating equation is obtained (lowercase letters denote logs):

$$q_{it} = \alpha_s l_{it} + \beta_s k_{it} + \phi_{it} \quad (2)$$

from which the firm-level log-TFP can subsequently be computed as the residual:

$$\hat{\phi}_{it} = q_{it} - \hat{\alpha}_s l_{it} - \hat{\beta}_s k_{it} \quad (3)$$

provided that consistent estimates of parameters α_{s} and β_{s} are available.

Equation (2) was estimated by ordinary least squares (LS), individual firm fixed effects (FE) and Levinsohn and Petrin (LP) methods to control for input-output simultaneity, (see Levinsohn and Petrin, 2003). The latter methodology is based on the idea that the error term in (2) can be decomposed into two components: $\phi_{ii} = \omega_{ii} + \varepsilon_{ii}$ where the first is observed by the firm but not by the econometrician while the second one is assumed to be an i.i.d noise uncorrelated with the inputs. To eliminate the correlation between ω_{ii} and input choice, LP propose using the following control function: $m_{it} = f_i(\omega_{it}, k_{it})$ where the demand of the intermediated inputs such as electricity, fuel and materials is considered a function of the productivity shock observed by the firm and the (predetermined) capital stock at time t. Under the assumption that the latter function is monotonic in m_{it} it is possible to invert $\omega_{it} = f_t^{-1}(m_{it}, k_{it})$ and substitute this expression into (2). The unobserved component of firm productivity causing the simultaneity bias is thereby eliminated and equation (2) can be estimated through non parametric methods after some additional moment restrictions. Unlike FE, this methodology does not assume that the unobserved productivity component is time invariant; in addition, it proposes a control function solidly based on profit maximization; finally, it is relatively easy to implement and less demanding in terms of data requirements than, for instance, the procedure in Olley and Pakes (1996).¹⁴

Given that the elasticity of output to capital and labor inputs may vary considerably across industries owing to intrinsic differences in production technologies, we run distinct regressions for each industry according to the two-digit SIC classification. In this way we can also control

¹⁴ Olley and Pakes (1996) propose the control function based on investment rather than on intermediate goods. The trouble is that this function cannot be inverted when investment is zero, a frequent occurrence in the data including our sample of manufacturing firms. For this reason we resort to the LP methodology.

for sector-specific time trends by introducing temporal fixed effects in the panel estimation procedure. To allow for some degree of firm heterogeneity within each two-digit SIC industry, fixed effects at the level of three-digit SIC codes were also included.

Firms with fewer than 5 employees were dropped from the sample prior to estimation for reasons of data reliability. Following the same line of reasoning, firms with outlier K/L ratios were also excluded, reducing the final sample to about 28,700 firms per year. Despite the trimming and quality controls, our sample is at least twice as large as those used in similar papers on Italian manufacturing firms.

Estimated labor and capital elasticities are shown in Table 5. Overall, the results obtained with the three estimation methods do not differ greatly, although the LS estimates exhibit slightly higher values compared with those derived with FE and LP, thus confirming the likely presence of the expected positive simultaneity bias. LP estimates show generally higher elasticities for the capital input and correspondingly lower estimates for the labor input compared with FE, the sum of the two coefficients attaining very close values in the two cases. Decreasing returns to scale (RTS) seem to be the prevalent regime in our estimates, although a formal test of constant RTS did not reject the null for the majority of sectors considered. There is a high correlation of estimated TFP levels across the three estimation methods, the Pearson correlation coefficient attaining values of 0.95 or higher.

6. Estimation of TFP differentials

Based on firm-level TFP estimates obtained with the procedure described above, we run the following regression:

$$\hat{\phi}_{it} = \delta UA + \eta ID + \rho f lagimp_{it} + \Sigma_h \mu_h firmsize_{it}^h + \gamma_g + \lambda_s + \sigma_t + \varepsilon_{it} \quad (4)$$

where

UA and ID are binary dummies indicating firms located in urban areas or industrial districts and δ and η are unknown coefficients measuring average TFP differentials between these two types of LLMAs and the remaining ones, which act as the reference group;

- *flagimp* is a control dummy signaling if L_{ii} has been either imputed or, alternatively, reported by the firm;
- firmsize^h is dummy variable taking the value of 1 if the size of the firm, measured by the number of employees, belongs to the *h*-th of *H* classes resulting from a discretization of the range of possible employment levels (size categories are: small firms ≤ 49 employees; medium-sized firms: 50-249; large firms: ≥ 250);
- γ_g , λ_s and ω_t are geographical area¹⁵(macro areas are: North-West, North-East, Centre; South), industry and year fixed effects;
- ε_{it} is an error term defined as the sum of two independent random components, an LLMA component and a purely idiosyncratic residual:

$$\varepsilon_{it} = \iota_r + \eta_{it} \quad (5) \,.$$

By including a firm size indicator in the specification we get rid of the differences in productivity levels that may depend on the fact that industrial districts can be more favorable areas for small business location (see Appendix 2 for a discussion on the relation between TFP and firm size). The geographical fixed effects γ_g allow for unobserved, time invariant factors affecting firm productivity across different areas. Industry fixed effects control for the influence that differences of sectoral composition between urban areas and industrial districts might have on the estimation results as well as for the well known problem of comparing productivity levels across different sectors.

Finally, the rationale for introducing a control for the data imputation process is to prevent any systematic bias possibly affecting our TFP estimates for firms with imputed employment levels from being transmitted to the estimates of spatial productivity differentials (which, in any event, would only occur if the share of imputed observations is not the same across urban areas, industrial districts and other LLMAs).

Given the assumptions about the error term in (5), we estimate equation (4) by clustering error terms at the individual LLMA level. Estimation results for this specification and for LP estimation method are reported in Table 6.¹⁶

¹⁵ Two broad partitions of the Italian territory are considered in this respect, corresponding, with some minor exceptions, to the NUTS1 and NUTS2 levels of the European regional classification.

The estimated TFP differential is positive and highly statistically significant for both urban areas and industrial districts. With respect to the reference group, a larger advantage is estimated for firms located in urban areas (10 per cent) than the gain observed for those based in industrial districts (3 per cent). In unreported evidence we show that these results do not change when TFP obtained through OLS or FE estimation methods is used. Moreover, we tested that the coefficients for urban areas and industrial districts are systematically different. For instance, for Model I in Table 6, we run a Wald test and obtain an F(1,688) with value 66.31 showing that we can strongly reject the null hypothesis that the two coefficients are not significantly different. This test is performed throughout all the other specifications and the results always reject this hypothesis. To save space we do not report the results of these tests.

In line with previous evidence, firms located in the Centre and, above all, in the South have much lower productivity levels than those in the North; the estimated gap is about 24 per cent for southern firms and 3 per cent for those located in the Centre.

Estimated coefficients display a significant non-linear relationship between firm size and log-TFP, suggesting that the productivity levels of medium-sized firms are only slightly higher than those of small firms, while large firms attain a greater advantage. However, the link between firm size and productivity may depend on the characteristics of the local environment. More precisely, we expect that small firms will achieve comparative advantages by locating in industrial districts. To explore this issue, we introduce into the regression the interaction between firm size and LLMA type (industrial district and urban area). This exercise shows that the productivity disadvantage of smaller firms is less marked within industrial districts. Overall, the estimates of the extra productivity in urban areas and industrial districts obtained with the baseline specification are confirmed.

The TFP advantages of urban areas and industrial districts diminish slightly when the three area dummies are replaced by a full set of fixed effects for Italy's 20 administrative regions (Table 6, Model III).

Our production function estimates do not take into account the so-called output price dispersion problem.¹⁷ Specifically, valued added has been deflated by a common industry-wide

¹⁶ For the sake of brevity, estimation results for the *flagimp* variable are not reported. In all cases, the estimated coefficients turned out to be negative and significant. This suggests that firms that do not report employment data are less productive than the average firm. In any event, including or excluding this control variable did not significantly alter our main estimation results.

¹⁷ See Del Gatto, Ottaviano and Pagnini (2008) for a discussion and a possible empirical solution.

price index. If firms in urban areas set higher prices, these would end up in the residual (see equation 3) and would bias the estimated urban productivity premium upwards. Prices in urban areas could be higher if firms located in them have more market power or produce higher quality goods, or because congestion or living costs are greater in cities than elsewhere. This potential criticism can be easily dismissed on the grounds that local conditions can hardly have a significant effect on the prices of manufacturing goods, whose relevant market is nation-wide if not global. Moreover, if local conditions have any effect, they would likely induce firms in cities to set lower prices, because competition is stiffer in cities or because the distributive sector is larger and more efficient than in other areas. Hence, our estimated urban productivity premium may be excessively conservative.

To check our results, we modify the definition of urban areas by using two different population thresholds (200,000 and 900,000) and rerun equation 4 accordingly. Notice that using the lower threshold would make interpreting the results more complex as now some industrial districts can also be included in the group of urban areas. In unreported evidence, we show that local productivity advantages in urban areas defined according to the two thresholds are a bit lower that those estimated in the baseline specification. For industrial districts, we obtain similar results with the 200,000 population threshold, but the productivity premium vanishes with the higher one (900,000 inhabitants). The latter result is due to the fact that the non-district, non-urban areas now encompass highly productive locations that are not at a disadvantage compared with industrial districts. Although we modify our definition of urban area as a robustness check, we want to emphasize that our previous choice of 500,000 inhabitants has the twofold advantage of generating a non-overlapping classification of the area types and at the same time setting a threshold above which it is likely that urbanization effects fully unfold.

We also run distinct regressions for each industry and unreported results indicate that comparative advantages associated with industrial districts and urban areas do not differ greatly across sectors. In Appendix 3 we report additional robustness checks, based on running similar regressions to equation (4) at aggregate rather than at individual firm level, using instrumental variables and for the subsample of small firms. These additional checks confirm our results.

7. Discussion of the main results and further extensions

One of the main results of our analysis is that firms located in urban areas outperformed those located in industrial districts in terms of productivity advantages. As a first step towards identifying the factors that may explain this finding, in this section we provide additional evidence on the evolution of the local productivity advantages, the role of the skill composition of the labor force and, finally, a quantile regression analysis of the data.

7.1 The evolution of local productivity advantages in urban areas and industrial districts

During the twelve years covered by our analysis, the Italian economy underwent some major transformations. The rapid and increasing diffusion of information and communication technologies, the upsurge of China, India and Brazil in world trade and finally the introduction of euro prompted a deep restructuring of Italian manufacturing firms. Moreover, all these factors probably gained momentum in the second part of the period: the euro was introduced in 2001, the effects of new technologies on workplace organization fed through in full in the same period and China's rise in world trade gained pace after the 2000. The question we want to investigate is whether these factors worked more in favor of firms located in urban areas or those established in industrial districts.

To this end, the three specifications considered in Table 6 were subsequently estimated by splitting the panel into two sub-periods, 1995 to 2000 and 2001 to 2006. The main findings point to a relative stability of the TFP advantage in urban areas over the two sub-periods, while the productivity premium estimated for industrial districts declined from about 4 per cent to 2 per cent, losing statistical significance when regional fixed effects are introduced (Table 7, Model III). These results suggest that firms located in urban areas coped with the shocks that hit the world economy better than district firms.

There are different possible explanations for this finding. Here we shall briefly indicate some. On one hand, it is possible that the larger endowment of skilled workers in urban areas permitted firms to make more efficient use of the new technologies and made it easier for them to improve product quality and introduce product innovation in response to increased competition (see the following sub-section). On the other, firms may have benefitted from the diversity of urban-area environment. In a period when the ability to update products and to innovate were crucial for firms' success, interactions with enterprises in other industries and with service firms amplified the possibility for firms under restructuring to undertake new production methods and test new strategies.¹⁸ The same advantages could not be reproduced by interactions within industrial districts, which involve small manufacturing firms belonging to relatively similar industries.

7.2 The role of human capital

A source of comparative advantage for cities may consist in higher human capital endowments. Urban areas may be especially successful in attracting more educated people because they allow skilled workers a better chance to find a good match with a firm on the large, diversified local job market. At the same time, cities may attract highly educated people through the local supply of urban-specific amenities. The empirical evidence detailing higher levels of labor force educational attainment in larger cities is clear and abundant. For the Italian case, recently Di Addario and Patacchini (2008) confirmed that skilled workers concentrate in the most populous cities and benefit from an urban wage premium.

If, ceteris paribus, firms located in urban workers hire more skilled workers than those in other local labor systems, omitting to control for the skill differential in the labor force will result in larger residuals in the estimated production function, which can be wrongly attributed to higher TFP levels.

In order to provide some new evidence on the impact of human capital on productivity in local labor markets, we relied on a measure of labor-force composition at firm level obtained from the Italian social security administration (Istituto Nazionale Previdenza Sociale, INPS) archives. The INPS database covers the entire universe of Italian firms with at least one employee and provides information on the total number of employees broken down into production and non-production workers, respectively defined as white-collar and blue-collar workers in what follows.

Using Italian data, Castellani and Giovannetti (2010) show that the share of blue-collar workers is strongly associated with firms' TFP, thus highlighting a possible misspecification in the production function. They suggest that the labor input should be split into different components capturing the different skill intensities, allowing for a more flexible specification of the production function.

¹⁸ The mechanism we are describing is similar to the "nursery effect" of Duranton and Puga (2001) with the difference that we do not think it can be restricted to young and small firms.

Building on this argument, we resort to a new set of production function estimates that include explicit controls for the labor force composition at the firm level. To do so, we pooled data on the number of blue- and white-collar workers from the INPS archives with our original Centrale dei Bilanci/Cerved (CEBI) database. The resulting panel includes fewer firms, owing to imperfect matching of firm codes in the two data sets and to a shorter time span covered by the INPS archive (it is limited to 1995-2002).

Using this database, we replicated our multi-step estimation strategy. In the first step the Levinsohn and Petrin method was employed to estimate the following augmented production function:

$$q_{it} = \alpha_s^b l_{it}^b + \alpha_s^w l_{it}^w + \beta_s k_{it} + \widetilde{\phi}_{it} \quad (2b)$$

where l^b and l^w and respectively denote (the log of) the number of blue- and white-collar employees. Subsequently, the revised TFP estimates obtained from the residuals of equation (2b) were used to run a TFP regression analysis akin to the one detailed in equations (4) and (5). Regression results based on TFP estimates derived from model (2b) are reported in Table 8. Considering that the augmented production function was estimated on a different sample, in order to provide a proper benchmark we also re-estimated TFP levels fitting the baseline Cobb-Douglas production function specification (equation 3) to the pooled INPS/CEBI data set. All in all, relying on a different panel of firms, featuring partially dissimilar employment data, does not appear to affect the estimation results substantially, as can be directly checked by comparing the results in Table 9 and Table 6.

Controlling for labor force composition, the estimated TFP advantage of firms located in urban areas remains large, with only a slight declining compared with the baseline results (from roughly 9 to 8 percentage points; see Tables 8 and 9). In other words, the productivity differential in favor of urban-area firms does not depend (or depends only marginally) on the fact that the labor force in urban areas has a larger share of skilled workers.

As a further refinement, we have obtained new estimates of the augmented production function specification, allowing the elasticity of output for the two labor inputs to take different values for firms located in industrial districts and urban areas. This less restrictive specification is introduced because white-collar workers may be more productive in urban areas, while blue-collar workers may be more efficiently employed in industrial districts. On the one hand, the growing literature on urban agglomeration has underlined the role of cities in the generation and transmission of new ideas that can spur innovation and productivity. Highly educated workers may be better equipped than less skilled ones to benefit from the flow of information that is diffused within urban areas by recurrent face-to-face interactions (Glaeser, Rosenthal and Strange, 2009; Glaeser and Ponzetto, 2010). On the other hand, the literature on industrial districts has emphasized the impact of agglomeration on skill accumulation by production workers, whose ability to "make things well" benefits from the local "industrial atmosphere" (according to Marshall's well-known definition), facilitating learning by doing.

Extended production function estimates (reported in Table 10) support the hypothesis that white-collar workers are more productive in urban areas: the estimated elasticity is higher for firms located in urban areas than for those located in districts or other areas, while blue-collar workers are more productive in non-urban areas. These results make sense in light of the theoretical a priori. However, the evidence that blue-collar workers are relatively more productive in industrial districts than in urban areas is not necessarily good news for districts' economic prospects: in the current competitive environment, marked by mounting competition from newly industrialized countries, the role of white-collar workers may prove crucial in fostering innovation and upgrading of firms' products.¹⁹

When different output elasticities to labor inputs are allowed for, estimated TFP differentials show a slight erosion of urban areas' productive advantage. Nonetheless, that advantage remains significant and substantial, ranging between 4.4 and 6.9 percentage points according to the different specifications (Table 11). The coefficient of the industrial district dummy now becomes not statistically significant, suggesting that the TFP differential in favor of industrial districts found by our baseline estimates may be basically due to the higher productivity of blue-collar workers in that environment rather than to a global shift in the efficiency of production. The large advantage of urban areas, instead, is due only in small part to the professional qualification of urban workers: in this sense, it remains unexplained.

7.3 Quantile analysis

The three sources of local productivity advantages discussed so far – agglomeration economies, selection and sorting – may have different effects when the TFP distributions of

¹⁹ See results in Section 7.1, the report on the recent evolution of Italian manufacturing sector by Brandolini and Bugamelli (2009) and also the discussion in Glaeser and Ponzetto (2010).

regions with a high and low concentration of economic activity are compared.²⁰ Agglomeration economies are assumed to raise the productivity of all the firms in a spatially concentrated region, so we should expect a rightward shift of the entire TFP distribution. Selection models show that the positive effect of geographical concentration mainly concerns the lower tail of the TFP distribution, since in dense areas less efficient firms are forced to exit the local market. Finally, if highly skilled workers or more efficient firms benefit to a greater extent from agglomeration economies we could have a positive effect on the upper tail of the TFP distribution (dilation effects).

To understand how these three forces shape the TFP distribution in urban areas and industrial districts we extend our econometric analysis to a quantile regression.²¹ This allows us to enrich the picture of the relationship between the response variable and the regressors at different points in the conditional distribution of the dependent variable., whereas the linear in mean regression analysis adopted so far simply assumes that the effect of the regressors on the conditional mean is representative of the shift in the entire distribution. Apart from this advantage, the quantile regression is also less exposed than the mean regression to the problems of outliers. Finally, being semiparametric in the sense that it avoids assumptions about the distributions of the error terms, this tool is particularly suitable for heteroskedastic data.

Results based on the quantile regression are reported in Table 12. To improve the efficiency of the estimation further, we bootstrapped the standard errors. In Figure 2 we also report a graphical analysis including OLS and the quantile regression coefficients for industrial districts and urban areas.

Several interesting patterns can be gleaned from this additional evidence. First, the productivity advantages associated with urban areas and industrial districts are confirmed across the different percentiles of the TFP distribution, thereby showing that the previous findings were not restricted to the impact of the covariates on the conditional mean. Moreover, apart from the first percentile of the TFP distribution, the productivity premium in urban areas is always greater than that observed in industrial districts, consistently with our previous results. Finally, the productivity advantages associated with industrial districts distinctly

²⁰ See Combes et al. (2009) for a thorough discussion of this topic.

²¹ For similar analyses, see Arimoto, Nakajima and Okazaki (2009) and Briant (2010). Quantile regression was proposed for the first time by Koenker and Bassett (1978).

diminish as we move from the lower to the upper tail of the distribution while the opposite holds true for urban areas.



Figure 2 – Quantile regression: ID and UA effects (1)

(1) The horizontal line corresponds to OLS coefficients, grey areas and horizontal dotted lines denote 95% confidence intervals for the estimated parameters. Quantiles vary by 0.05, from 0.05 to 0.95.

The fact that industrial district and especially urban area coefficients are positive and significant at all quantiles suggests that agglomeration economies actually shift the entire TFP distribution rightward or that at least part of them positively affect TFP of all the firms located in dense regions. At the same time, the magnitude of these coefficients vary across quantiles and according to different patterns in industrial districts and urban areas, indicating that selection and sorting may also help shape the TFP distribution.

We also detected some evidence consistent with selection in industrial districts, as their productivity advantages are stronger in the lower tail of the TFP distribution, while we do not find a similar effect in urban areas (Figure 2, panels b and c). The alternative interpretation based on sorting would argue that more efficient firms reap greater benefits from the kind of agglomeration economies created in urban areas, while, conversely, less efficient firms are better equipped to gain from the externalities generated in industrial districts.

Although disentangling these alternative explanations is beyond the scope of the paper, we doubt that selection models can actually help explain the different patterns in the relationship between geographical concentration and TFP found in industrial districts and urban areas: the geographical scale of LLMAs adopted in our approach is too small and therefore not suited to represent the relevant market in the case of manufacturing products: Accetturo et al. (2011) show that a selection effect may emerge when a broader spatial scale is considered.

Hence, we conjecture that the differences between the two curves in Figures 2b and 2c, may be driven by some kind of sorting process. Industrial districts are able to generate local productivity advantages that can be more effectively appropriated by less efficient firms, while the externalities produced in urban areas can be better exploited by the more efficient firms. The dense network of productive relationships generated by the industrial districts seems to bring the greatest benefits to firms that would be very inefficient if they operated on their own outside that network. As for urban areas, more efficient firms that are able to survive in that tougher competitive environment are also likely to be the ones to benefit more from interaction with firms in other sectors (diversity) or from improved matching with a skilled local labor force.²² This result partially contradicts the finding of Duranton and Puga (2001) that cities benefit younger and relatively less efficient firms. Nor does it square fully with the evidence presented by Holmes and Stevens (2010) showing that US cities attract small, less efficient firms producing high-quality goods.

Plainly, further investigation of these results is needed in order to check their robustness and for possible policy prescriptions.

8. Conclusion

This paper has investigated the issue of local productivity advantages, using data on 29,000 Italian manufacturing firms over an observation period of 12 years (1995-2006). We

²² For a model where agglomeration benefits are higher for more productive firms, see Combes et al (2009). For a model taking into account the effects of an intense competition, see Venables (2011).

mapped firms into three non-overlapping categories according to their respective location (urban areas, industrial districts, and other areas) and performed firm-level TFP estimates using a broad set of techniques.

On the whole, our analysis suggests that spatial concentration exerts favorable effects on local productivity. The estimated coefficients for the urban areas and industrial districts dummies are both positive and significant. However, locating in an urban areas is generally more beneficial than in an industrial district (with an estimated coefficient 3 to 5 times greater depending on the specification used).

While manufacturing firms located in urban areas on average employ a more skilled labor force, TFP estimates that explicitly control for this skill differential show that the productivity advantage of large cities depends to only a minor extent on differences in the human capital endowment of employees. Using quantile regression techniques, we also show that industrial districts generate local productivity advantages that can be more effectively appropriated by less efficient firms, while the externalities arising within cities can be better exploited by more efficient enterprises.

With the purpose of evaluating the dynamic pattern of productivity over the period (1995-2006), we run a new regression analysis splitting the sample in two sub-periods. It turns out that comparative advantages of urban areas remain stable while those of industrial districts show a tendency to decline over time. Within cities agglomeration economies have remained vital, even in a period characterized by the growing globalization. Thus, our results suggest that firms operating within urban areas, better than those located in industrial districts, have shown a high degree of resilience to the shocks that hit the world economy over the last decade.

Finally, let us consider the policy implications of our findings. The purpose of our work is not to suggest a path of development for the local productive systems, nor do we intend to propose active policy measures to support industrial districts or urban areas, since defining optimal policies for clusters is very difficult. In fact, the question is: what should these policies do? If the answer is "solve major inefficiencies", then policy makers should know exactly where these inefficiencies come from (see Duranton, 2011, for a more detailed analysis). Appropriate policies are thus not easy to formulate and must be supported by a thorough knowledge of the inefficiencies to be addressed (linked, for example, to congestion or to the fading away of positive externalities). Such knowledge is a prerequisite for effective policy measures. The present paper can be considered as a first step in this line of research.

TABLES

Table 1

				,,	
Authors	Stra	tegy	Model	Variable of interest	Agglomeration advantage (1)
Signorini (1994)	Firms' performance	Case study on Prato industrial district (1982-1989)	OLS	Total factor productivity	+ 41 %
Fabiani et al. (2000) (2)	Firms' performance	Cross section in 1995	Descriptive stats.	ROI	+ 2.0 p.p.
		(firm level)	&	ROE	+ 4.1 p.p.
			Stochastic Frontiers – ML estimates	Value added per worker	+ 1.3 %
Gola and Mori (2000)	Export structure	Panel data (firm level) 1,092 obs period 1983-1995	Fixed effect estimates	Normalized trade balance	+ 3.4 %
Bronzini (2000)	Export performance	Data at provincial level; pooled data (1995-1997)	OLS, SURE estimates	Export propensity (log of export per worker as a share of national average)	+ 7.0 %
Becchetti and Rossi (2000)	Export intensity	Mediocredito survey. Firm level data: 1989-1991	Tobit estimates	Share of export on total sales	+ 3.6
		(avg) 3,695 obs.	Probit estimates	Exporter status (marginal effect)	+ 20 %
Cainelli and De Liso (2005)	Firms performance	Period 1992-1995 (2,821 obs.)	OLS, IV estimates	Rate of change of real value added	+ 2.0 / 2.6 %
Becchetti and Castelli (2005)	Firms' performance	Mediocredito Survey (two waves: 1995- 1997 and 1998- 2000)	Stochastic Frontier	Value added per worker	+ 1.8 %
Bugamelli and Infante (2005)	Exporter status	Firm level (1982-1999).	Probit estimates	Exporter status	+ 0.02
Cingano and Schivardi (2005) (3)	Firms' performance	Firm level (1,602 obs.)	OLS estimates	Average TFP growth rate 1986-1998	+ 0.2 / 0.5 %
Foresti, Guelpa and Trenti (2009)	Firms' performance	Different indicators 2006	Descriptive statistics	ROI	- 0.25 p.p.

The importance of being agglomerated: the district effect in Italy

(1) Firms in districts with respect to firms not in districts. - (2) The authors also perform a sectoral analysis of firms' efficiency using the stochastic frontier approach, finding evidence of lower for firms located in districts for 8 out of 13 sectors. - (3) They produce indirect (though robust) evidence in favor of a district effect, testing whether the increase of the industry degree of specialization in LLMAs (an index of externality typical of districts) determines a change in TFP growth.

The	sample: number o	of observations		
Sectors	Industrial districts	Urban areas	Other	Total
Food products, beverages and tobacco	9,985	4,837	10,549	25,371
Textiles and textile products	28,656	6,418	7,528	42,602
Leather and leather products	11,847	3,456	2,078	17,381
Wood and products of wood and cork (except furniture)	5,588	1,575	3,898	11,061
Pulp, paper and paper products; recorded media; printing services	9,046	10,048	4,934	24,028
Coke, refined petroleum products and nuclear fuel	290	496	562	1,348
Chemicals, chemical products and man-made fibres	4,938	5,810	2,796	13,544
Rubber and plastic products	11,512	5,152	5,275	21,939
Other on metallic mineral products	10,266	3,205	8,435	21,906
Basic metals and fabricated metal products	40,834	18,479	20,952	80,265
Machinery and equipment n.e.c.	29,635	14,547	12,286	56,468
Electrical and optical equipment	14.387	12.741	7.540	34.668
Transport equipment	3 658	3,725	3 759	11.142
Other manufactured goods n.e.c.	18.371	5,690	7,090	31.151
	,	0,000	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	01,101
North-West	80,260	52,260	27,198	159,718
North-East	74,113	18,268	28,630	121,011
Centre	40,088	14,566	16,580	71,234
South and islands	4,552	11,085	25,274	40,911
1995-2000	93,251	46,803	43,783	183,837
2001-2006	105,762	49,376	53,899	209,037
Total	199,013	96,179	97,682	392,874

Sectors		Size (average)		Size (median)			
-	Industrial districts	Urban areas	Other	Industrial districts	Urban areas	Other	
Food products, beverages and tobacco	53.1	95.4	46.0	19.1	21.5	17.2	
Textiles and textile products	44.9	43.8	68.2	20.0	15.9	20.0	
Leather and leather products	35.1	32.7	48.2	18.0	17.8	18.6	
Wood and products of wood and cork (except furniture)	27.8	25.7	28.9	17.6	13.5	14.3	
Pulp, paper and paper products; recorded media; printing services	37.4	57.0	44.0	16.0	14.5	16.1	
Coke, refined petroleum products and nuclear fuel	93.7	276.6	39.4	19.0	34.0	14.0	
Chemicals, chemical products and man-made fibres	62.8	154.9	87.2	21.0	40.0	19.0	
Rubber and plastic products	42.2	77.9	51.0	21.3	21.0	21.1	
Other on metallic mineral products	59.3	52.3	36.3	20.0	19.0	16.0	
Basic metals and fabricated metal products	36.2	45.5	36.3	16.8	14.8	16.7	
Machinery and equipment n.e.c.	47.8	67.2	80.6	19.8	18.5	19.5	
Electrical and optical equipment	47.7	92.0	65.1	17.5	17.0	16.3	
Transport equipment	104.4	329.6	149.2	23.2	26.0	23.1	
Other manufactured goods n.e.c.	34.1	29.1	33.6	17.0	14.4	16.9	
North-West	49.0	93.0	60.7	19.1	18.7	18.6	
North-East	45.3	51.0	61.9	19.0	18.2	19.2	
Centre	32.0	75.7	52.5	16.3	14.7	16.3	
South and islands	38.9	50.8	40.2	19.8	15.4	15.7	
1995-2000	46.0	88.0	60.6	20.0	18.8	19.6	
2001-2006	42.2	67.6	49.3	17.1	16.2	16.0	
Total	43.9	77.5	54.4	18.4	17.3	17.5	

Descriptive statistics: Firms' size (number of employees)

Sectors	Added value per worker (average)			Added value per worker (median)			
	Industrial districts	Urban areas	Other	Industrial districts	Urban areas	Other	
Food products, beverages and tobacco	64.5	66.7	57.0	159.2	157.2	172.4	
Textiles and textile products	43.1	43.5	35.5	73.1	61.1	62.4	
Leather and leather products	41.9	41.4	36.6	46.7	33.9	39.5	
Wood and products of wood and cork (except furniture)	41.9	44.7	38.9	74.3	69.4	74.8	
Pulp, paper and paper products; recorded media; printing services	52.0	55.1	48.9	99.3	81.9	101.9	
Coke, refined petroleum products and nuclear fuel	118.5	111.9	92.7	253.4	425.5	224.4	
Chemicals, chemical products and man-made fibres	69.6	74.8	66.1	137.2	134.0	141.4	
Rubber and plastic products	48.4	49.8	44.6	95.6	96.6	104.2	
Other on metallic mineral products	55.2	54.4	50.8	118.1	127.5	139.2	
Basic metals and fabricated metal products	51.2	51.7	45.5	80.3	74.6	68.9	
Machinery and equipment n.e.c.	53.1	54.8	50.1	60.6	57.1	60.5	
Electrical and optical equipment	50.4	54.8	47.0	52.9	51.4	51.9	
Transport equipment	46.0	48.3	42.8	70.2	72.0	68.3	
Other manufactured goods n.e.c.	39.6	45.0	39.8	57.3	61.5	62.9	
North-West	52.2	56.6	51.3	88.2	78.9	88.7	
North-East	50.2	52.6	49.7	77.8	72.2	80.9	
Centre	44.6	52.9	46.0	63.9	77.1	80.6	
South and islands	38.3	43.8	40.5	78.3	91.5	103.9	
1995-2000	44.9	48.4	42.2	77.1	76.8	88.0	
2001-2006	53.7	58.9	51.1	81.1	80.8	89.7	
Total	49.6	53.8	47.1	79.2	78.8	89.0	

Descriptive statistics: Added value per worker (thousands of euros)

(standard errors in brackets)									
	Levi	nsohn-Petri	n	Fix	ed effects	6	Ord	Ordinary least squares	
Sectors	Labor coeff.	Capital coeff.	RTS	Labor coeff.	Capital	RTS	Labor coeff.	Capital coeff.	RTS
Food products,									
beverages and									
tobacco	0.572 (0.013)	0.218 (0.030)	0.790	0.673 (0.010)	0.200	0.873	0.837 (0.005)	0.195 (0.004)	1.032
Textiles and textile	()	()		()	(/		(/	()	
products	0.708	0.272	0.980	0.866	0.131	0.997	0.871	0.123	0.993
•	(0.008)	(0.015)		(0.008)	(0.007)		(0.004)	(0.003)	
Leather and leather	()	(0.0.0)		(00000)	(0.000)		(0.000)	(0.000)	
products	0.716	0.261	0.977	0.842	0.136	0.978	0.884	0.137	1.02
•	(0.009)	(0.020)		(0.011)	(0.009)		(0.005)	(0.004)	
Wood and products	(()		()	()		()	(
of wood and cork									
(except furniture)	0.724	0.235	0.959	0.830	0.110	0.940	0.898	0.125	1.023
()	(0.018)	(0.027)		(0.012)	(0,009)		(0,006)	(0, 004)	
Pulp, paper and	(0.0.0)	(0.02.)		(01012)	(0.000)		(0.000)	(0.00.)	
naper products:									
recorded media:									
printing services	0.710	0.195	0.905	0.744	0.148	0.893	0.907	0.133	1.04
printing conviceo	(0.016)	(0.015)	0.000	(0.010)	(0.008)	0.000	(0.005)	(0.003)	1.0 1
Coke refined	(0.010)	(0.010)		(0.010)	(0.000)		(0.000)	(0.003)	
netroleum products									
and nuclear fuel	0 519	0 557	1 076	0 569	0 242	0.811	0 851	0 219	1.06
	(0.087)	(0.102)	1.070	(0.041)	(0.042)	0.011	(0.023)	(0.016)	1.00
Chemicals chemical	(0.007)	(0.102)		(0.041)	(0.042)		(0.023)	(0.010)	
products and man-									
made fibres	0.660	0 292	0 952	0 750	0 171	0 921	0 925	0 114	1.03
	(0.018)	(0.020)	0.002	(0.013)	(0.012)	0.021	(0.020	(0.005)	1.00
Rubber and plastic	(0.010)	(0.023)		(0.013)	(0.012)		(0.007)	(0.003)	
nroducte	0 606	0.284	0.081	0 701	0 166	0 057	0 855	0 171	1 0 2
producis	(0.012)	(0.010)	0.301	(0.008)	(0.008)	0.337	(0.005)	(0.002)	1.02
Other nen metellie	(0.012)	(0.019)		(0.008)	(0.008)		(0.005)	(0.003)	
minoral products	0 665	0 212	0.077	0.916	0 121	0.046	0 000	0 171	1 05
mineral products	0.005	0.312	0.977	(0.000)	(0.000)	0.940	0.000	0.171	1.05
Decia motole and	(0.012)	(0.031)		(0.009)	(0.009)		(0.005)	(0.003)	
Basic metals and									
tabricated metal	0 707	0.007	0.004	0.004	0 4 0 7	0.040	0.074	0.400	4.04
products	0.727	0.207	0.934	0.821	0.127	0.948	0.071	0.139	1.01
Mashinanyand	(0.004)	(0.007)		(0.004)	(0.003)		(0.002)	(0.001)	
iviachinery and	0 707	0.010	0.040	0.004	0.405	0.000	0.040	0.400	4.04
equipment n.e.c.	0.737	0.212	0.949	0.831	0.135	0.966	0.912	0.102	1.01
Electrical en la dist	(0.007)	(0.011)		(0.005)	(0.004)		(0.003)	(0.002)	
Electrical and optical	0 700	0.400	0.000	0.005	0.440	0.0.15	0.004	0.440	4.0.1
equipment	0.730	0.193	0.923	0.825	0.119	0.945	0.904	0.110	1.014
	(0.008)	(0.012)	- ·	(0.007)	(0.006)		(0.004)	(0.003)	
Transport equipment	0.758	0.196	0.954	0.873	0.110	0.983	0.911	0.096	1.00
	(0.015)	(0.019)		(0.013)	(0.010)		(0.006)	(0.004)	
Other manufactured									
goods n.e.c.	0.746	0.210	0.956	0.856	0.139	0.995	0.935	0.107	1.04
	(0.009)	(0.015)		(0.008)	(0.007)		(0.004)	(0.003)	

(standard errors in brackets) (2)						
Model I	Model II	Model III (3)				
0.102***	0.108***	0.092***				
(0.01)	(0.01)	(0.01)				
0.029***	0.036***	0.016*				
(0.01)	(0.01)	(0.01)				
0.033***		0.037***				
(0.01)		(0.01)				
0.160***		0.164***				
(0.01)		(0.01)				
-0.001	-0.001					
(0.01)	(0.01)					
-0.035**	-0.036**					
(0.01)	(0.01)					
-0.242***	-0.242***					
(0.01)	(0.01)					
	-0.039*					
	(0.02)					
	0.030					
	(0.03)					
	-0.037**					
	(0.01)					
	-0.001					
	(0.03)					
344,353	344,353	344,353				
0.677	0.678	0.679				
	Model I 0.102*** (0.01) 0.029*** (0.01) 0.033*** (0.01) 0.160*** (0.01) -0.001 (0.01) -0.035** (0.01) -0.035** (0.01) -0.242*** (0.01) -0.242*** (0.01) -0.242*** (0.01)	Model I Model II 0.102*** 0.108*** (0.01) (0.01) 0.029*** 0.036*** (0.01) (0.01) 0.033*** (0.01) (0.01) (0.01) 0.160*** (0.01) (0.01) -0.001 0.160*** (0.01) (0.01) -0.001 -0.001 -0.001 (0.01) (0.01) -0.035** -0.036** (0.01) (0.01) -0.242*** -0.242*** (0.01) (0.01) -0.039* (0.02) 0.030 (0.03) -0.037** (0.01) -0.037** (0.01) -0.037** (0.01) -0.001 (0.03) 344,353 344,353 0.677 0.678				

Estimation results on firm-level data. Domondont

Sources: Based on Centrale dei Bilanci and Cerved data. (1) All specifications include year and industry fixed effects plus a control for imputed employees data. - (2) Standard errors are corrected for clustering at the level of individual LLMAs. - (3) Includes 20 region fixed effects.

	Мо	del I	Мос	del II	Мос	lel III
	1995-2000	2001-2006	1995-2000	2001-2006	1995-2000	2001-2006
UA	0.103***	0.102***	0.112***	0.105***	0.094***	0.090***
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
ID	0.038***	0.021*	0.048***	0.025**	0.023**	0.010
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Medium-sized	0.011	0.053***			0.016*	0.056***
	(0.01)	(0.01)			(0.01)	(0.01)
Large	0.133***	0.187***			0.140***	0.190***
	(0.01)	(0.02)			(0.01)	(0.02)
North-East	-0.002	0.000	-0.002	-0.000		
	(0.01)	(0.01)	(0.01)	(0.01)		
Centre	-0.032	-0.039***	-0.032	-0.039***		
	(0.02)	(0.01)	(0.02)	(0.01)		
South	-0.267***	-0.220***	-0.267***	-0.220***		
	(0.01)	(0.01)	(0.01)	(0.01)		
UA*medium- sized			-0.051**	-0.029		
			(0.02)	(0.02)		
UA*large			0.010	0.052		
			(0.03)	(0.04)		
ID*medium- sized			-0.047***	-0.031*		
			(0.01)	(0.01)		
ID*large			-0.022	0.017		
			(0.03)	(0.04)		
Number of obs.	166,168	178,185	166,168	178,185	166,168	178,185
Adjusted R ²	0.690	0.666	0.690	0.667	0.692	0.668

Estimation results on firm-level data, by period. Dependent variable: log of TFP measured through LP method (1) (standard errors in brackets) (2)

Sources: Based on Centrale dei Bilanci and Cerved data. (1) All specifications include year and industry fixed effects plus a control for imputed employees data. - (2) Standard errors are corrected for clustering at the level of individual LLMAs. - (3) Includes 20 region fixed effects.

Estimation results on firm-level data, using two labor inputs (white-collar and blue-collar workers). Dependent variable: log of TFP measured through LP method (1) (standard errors in brackets) (2)

	Model I	Model II	Model III (3)
UA	0.078***	0.078***	0.069***
	(0.01)	(0.01)	(0.01)
ID	0.026**	0.040***	0.014
	(0.01)	(0.01)	(0.01)
Medium-sized	0.133***		0.137***
	(0.01)		(0.01)
Large	0.336***		0.344***
	(0.02)		(0.02)
North-East	0.018	0.019	
	(0.01)	(0.01)	
Centre	-0.012	-0.013	
	(0.02)	(0.02)	
South	-0.237***	-0.236***	
	(0.02)	(0.02)	
UA*medium-sized		-0.013	
		(0.02)	
UA*large		0.062	
		(0.04)	
ID*medium-sized		-0.057***	
		(0.01)	
ID*large		-0.060	
		(0.04)	
Number of observations	188,275	188,275	188,275
Adjusted R ²	0.796	0.796	0.797

Sources: Based on Centrale dei Bilanci, Cerved aand INPS data. (1) All specifications include year and industry fixed effects plus a control for imputed employees. Data on labor inputs are drawn by INPS dataset. Estimation period: 1995-2002. - (2) Standard errors are corrected for clustering at the level of individual LLMAs. - (3) Includes 20 regional fixed effects.

Estimation results on firm-level data, using only one labor input (white + blue-collar workers). Dependent variable: log of TFP measured through LP method (1) (standard errors in brackets) (2)

	Model I	Model II	Model III (3)
UA	0.089***	0.089***	0.079***
	(0.01)	(0.01)	(0.01)
ID	0.033**	0.046***	0.020
	(0.01)	(0.01)	(0.01)
Medium-sized	0.130***		0.135***
	(0.01)		(0.01)
Large	0.322***		0.330***
	(0.02)		(0.02)
North-East	0.015	0.015	
	(0.01)	(0.01)	
Centre	-0.026	-0.027	
	(0.02)	(0.02)	
South	-0.260***	-0.259***	
	(0.01)	(0.01)	
UA*medium-sized		-0.014	
		(0.02)	
UA*large		0.051	
		(0.03)	
ID*medium-sized		-0.053***	
		(0.01)	
ID*large		-0.047	
		(0.04)	
Number of observations	188,275	188,275	188,275
Adjusted R ²	0.801	0.801	0.803

Sources: Based on Centrale dei Bilanci, Cerved aand INPS data. (1) All specifications include year and industry fixed effects plus a control for imputed employees. We use only one labor input drawn by INPS dataset (White + Blue collars). Estimation period: 1995-2002. - (2) Standard errors are corrected for clustering at the level of individual LLMAs. - (3) Includes 20 regional fixed effects.

						Canital	
	White-collar workers Blue-collar workers						Capital
Area Type	Non-ID/UA	Urban areas (1)	Industrial districts (1)	Non-ID/UA	Urban areas (1)	Industrial districts (1)	-
Sectors							-
Food products, beverages and tobacco	0,200***	0,043	-0,030	0,256***	-0,023	0,045***	0,264***
	(0,020)	(0,029)	(0,024)	(0,018)	(0,021)	(0,016)	(0,064)
Textiles and textile products	0,182***	0,094***	-0,005	0,355***	-0,057***	0,012	0,407***
	(0,019)	(0,024)	(0,016)	(0,013)	(0,017)	(0,010)	(0,027)
Leather and leather products	0,139***	0,038	0,013	0,398***	0,014	0,008	0,376***
	(0,034)	(0,038)	(0,033)	(0,021)	(0,017)	(0,016)	(0,030)
Wood and products of wood and cork (except furniture)	0,208***	0,020	-0,026	0,408***	0,010	0,017	0,349***
	(0,024)	(0,039)	(0,027)	(0,020)	(0,018)	(0,014)	(0,029)
Pulp. paper and paper products; recorded media: print. services	0,193***	0,055**	-0,052**	0,263***	-0,017	0,049***	0,269***
	(0,021)	(0,026)	(0,024)	(0,021)	(0,016)	(0,014)	(0,024)
Coke. refined petroleum products and nuclear fuel	0,025	-0,009	0,143	0,124***	0,025	-0,084	0,886***
	(0,087)	(0,143)	(0,152)	(0,077)	(0,107)	(0,139)	(0,195)
Chemicals. chemical products and man- made fibers	0,278***	0,105***	0,014	0,167***	-0,091***	-0,006	0,495***
	(0,030)	(0,028)	(0,033)	(0,024)	(0,028)	(0,029)	(0,039)
Rubber and plastic products	0,169***	0,067***	0,021	0,362***	-0,024	0,008	0,439***
	(0,020)	(0,027)	(0,023)	(0,021)	(0,017)	(0,014)	(0,028)
Other non metallic mineral products	0,125	0,063	-0,004	0,312***	-0,020	0,019	0,505***
	(0,018)	(0,034)	(0,021)	(0,022)	(0,021)	(0,013)	(0,026)
Basic metals and fabricated metal products	0,167***	0,042***	0,018	0,413***	-0,017	-0,002	0,301***
	(0,008)	(0,016)	(0,010)	(0,008)	(0,009)	(0,006)	(0,015)
Machinery and equipment n.e.c.	0,251***	0,086***	0,008	0,324***	-0,059***	0,000	0,319***
	(0,013)	(0,016)	(0,013)	(0,011)	(0,012)	(0,010)	(0,014)
Electrical and optical equipment	0,275***	0,082***	0,007	0,266***	-0,062***	0,001	0,358***
	(0,017)	(0,017)	(0,019)	(0,015)	(0,013)	(0,014)	(0,027)
Transport equipment	0,221***	0,050	-0,066	0,313***	-0,042	0,037	0,335***
Other menufacture d	(0,031)	(0,062)	(0,035)	(0,033)	(0,042)	(0,023)	(0,061)
goods n.e.c.	0,160***	0,061	0,025	0,404***	-0,026	-0,023	0,291***
	(0,025)	(0,035)		(0.020)	(0,023)	(0,020)	(0,028)

Production function coefficients by area type, industry and labor force characteristics (standard errors in brackets)

Sources: Based on Centrale dei Bilanci, Cerved and INPS data.

(1) Deviations from Non-ID/UA coefficients.

Estimation results on firm-level data, using two labor inputs (white- and blue-collar workers) and two distinct coefficients for UA and ID. Dependent variable: log of TFP measured through LP method (1) (standard errors in brackets) (2)

	Model I	Model II	Model III (3)
UA	0.053***	0.068***	0.044***
	(0.01)	(0.01)	(0.01)
ID	0.002	0.020	-0.010
	(0.01)	(0.01)	(0.01)
Medium-sized	0.126***		0.131***
	(0.01)		(0.01)
Large	0.317***		0.324***
	(0.01)		(0.01)
North-East	0.019	0.019	
	(0.01)	(0.01)	
Centre	-0.012	-0.013	
	(0.02)	(0.02)	
South	-0.233***	-0.232***	
	(0.01)	(0.01)	
UA*medium-sized		-0.065***	
		(0.02)	
UA*large		-0.039	
		(0.04)	
ID*medium-sized		-0.073***	
		(0.01)	
ID*large		-0.083*	
		(0.04)	
Number of observations	188,275	188,275	188,275
Adjusted R ²	0.800	0.800	0.801

Sources: Based on Centrale dei Bilanci, Cerved aand INPS data.

(1) We use two labor inputs drawn from the INPS dataset (white- and blue-collar workers) and two distinct coefficients for industrial districts and urban areas. All specifications include year and industry fixed effects plus a control for imputed employees. Estimation period: 1995-2002. - (2) Standard errors are corrected for clustering at the level of individual LLMAs. - (3) Includes 20 regional fixed effects.

	Q01	Q <i>05</i>	Q10	Q25	Q50	Q75	Q90	Q95	Q99
UA	0.069**	0.078***	0.077***	0.083***	0.092***	0.111***	0.123***	0.128***	0.160***
	(0.02)	(0.01)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.01)
ID	0.114***	0.061***	0.041***	0.027***	0.020***	0.018***	0.020***	0.015***	0.023**
	(0.02)	(0.01)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.01)
Medium-	0.117***	0.068***	0.064***	0.052***	0.037***	0.019***	0.001	-0.011**	-0.043***
sized	(0.02)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.01)
Large	0.185***	0.135***	0.141***	0.153***	0.155***	0.162***	0.184***	0.186***	0.192***
	(0.05)	(0.01)	(0.01)	(0.00)	(0.00)	(0.00)	(0.01)	(0.01)	(0.02)
North-East	0.062*** <i>(0.02)</i>	0.010* <i>(0.00)</i>	0.002 <i>(0.00)</i>	-0.001 <i>(0.00)</i>	-0.003* <i>(0.00)</i>	-0.007*** <i>(0.00)</i>	-0.014*** <i>(0.00)</i>	-0.012*** <i>(0.00)</i>	-0.028*** (0.01)
Centre	-0.140*** <i>(0.03)</i>	-0.070*** (0.01)	-0.056*** <i>(0.00)</i>	-0.046*** <i>(0.00)</i>	-0.039*** <i>(0.00)</i>	-0.034*** <i>(0.00)</i>	-0.024*** <i>(0.00)</i>	-0.021*** <i>(0.01)</i>	-0.035*** <i>(0.01)</i>
South	-0.640*** <i>(0.04)</i>	-0.404*** (0.01)	-0.317*** <i>(0.01)</i>	-0.249*** <i>(0.00)</i>	-0.219*** <i>(0.00)</i>	-0.202*** (0.00)	-0.196*** <i>(0.00)</i>	-0.190*** <i>(0.01)</i>	-0.156*** <i>(0.01)</i>
N. obs. <i>Pseudo R</i> ²	344353 0.2728	344353 0.4712	344353 0.5137	344353 0.5107	344353 0.4722	344353 0.4212	344353 0.3989	344353 0.3933	344353 0.3856

Quantile regression. Estimation results on firm-level data. Dependent variable: log of TFP measured through LP method (1) (2) (standard errors in brackets) (3)

(1) All specifications include year and industry fixed effects plus a control for imputed employees data. - (2) Q01, ...,Q99 indicate estimation carried out at the different percentiles of the TFP distribution (Q01 denotes the first percentile ,and so on). - (3) Bootstrapped standard errors, 20 replications.

Appendix 1 - Imputing employee data

Average unit labor costs measured on the sub-sample of firms for which employment counts information is available provide the information needed to recover missing labor input data. To allow for possible heterogeneity in mean wages, the sample was stratified according to a number of relevant firm characteristics.

In particular, mean wages are allowed to vary across sector, geographical area and type of local labor market. Additional firm-level wage heterogeneity is also controlled for by stratifying the sample according to firm size, measured by value added, and profitability. Larger firms may have a different skill composition of the labor force, and consequently different mean wages. At the same time, more profitable firms are more likely to pay wage premiums, thus sustaining higher total labor costs for a given number of employees.

In each stratum the median of observed firm-level average labor costs was computed, and these estimates were subsequently used to impute missing employment data by taking the ratio of total firm labor costs to the median wage of the stratum in which the firm is classified.

Appendix 2 - The relation between TFP and firm size

Estimates of agglomeration effects on TFP levels discussed so far were based on regression analyses at the firm level. As such, they tend to be prone to measurement problems and the presence of outliers, possibly affecting estimation results in unexpected ways.

Considering that no constraints on returns to scale were introduced when estimating the production function at the firm level, the introduction of a relationship between estimated TFP levels and firm size can be motivated by the existence of a possibly non (log)linear function linking TFP to firm size. To illustrate the argument, let us assume that the log TFP level can be expressed as a generic function of firm size, measured by the employment level,

 $\phi_{it} = h(l_{it}) \quad (1a).$

Under the hypothesis that the function h(.) can be well approximated by means of a polynomial of order p, equation (2) in the main text can be restated as:

$$q_{it} = \alpha \ l_{it} + \beta \ k_{it} + \rho_0 + \rho_1 l_{it} + \rho_2 l_{it}^2 + \dots + \rho_1 l_{it}^p$$

= $\widetilde{\alpha} \ l_{it} + \beta \ k_{it} + \widetilde{\phi}_{it}$ (2a)

where $\widetilde{\alpha} = \alpha + \rho_1$ and $\widetilde{\phi}_{it} = \rho_0 + \rho_2 l_{it}^2 + ... + \rho_1 l_{it}^p$.

Expression (2*a*) above shows how estimating a Cobb-Douglas production function with unrestricted elasticities purges the residual TFP estimates of scale effects only under the restrictive assumption of an exact log-linear relation between individual TFP and firm size. In the presence of a more general non linear relation, production function residuals will be correlated with higher powers of the labor input.²³

As a consequence, omitting to control for firm size in equation (4) in the main text may result in biased estimates of agglomeration productivity advantages if size is uneven across different LLMA classes, (i.e., if the UA and ID regressors are correlated with firm size).

Appendix 3 - Additional robustness checks

In this section we discuss robustness checks based on running similar regressions to equation (4) in the main text at aggregate rather than at individual firm level, using instrumental variables and for the subsample of small firms.

Considering that the research focuses on productivity differentials at the level of local labor markets, a more robust estimation approach can be implemented if individual TFP levels are aggregated prior to running the regression analysis. For this purpose, data were first aggregated at the level of the industry/LLMA/year by taking employment-weighted averages of individual TFP levels, the choice of the weighting variable being motivated by the expectation that data quality deteriorates as firm size decreases:

$$\phi_{srt} = \frac{1}{L_{srt}} \sum_{i \in (r,s,t)} L_{it} \phi_{it} \quad (3a)$$

²³ The correlation between inputs and the residual term stemming from equation (2*a*) when p>1 provides an additional argument in favor of estimation methods that can cope with this issue, like the Olley-Pakes (1996) and Levinshon-Petrin (2003) procedures.

Using data at this level of aggregation, equation (4) in the main text was re-estimated by weighted least squares, using the number of firms in each stratum as weight. Estimation results, displayed in Table a1, while confirming the previous evidence of a productivity surplus in urban areas and industrial districts, also show a larger differential, especially in favor of urban areas, where it rises to about 17 per cent. Introducing unobserved regional effects lowers the estimated comparative advantages for urban areas and industrial districts, as in the previous section (see Table a1, column 2).

At this stage, a first attempt was made to deal with the endogeneity issue that is likely to affect the variables identifying urban areas and industrial districts with respect to local productivity levels. In fact, since firm location is not set exogenously but results from individual optimizing choices, plant location can be correlated with unobserved firm characteristics and, in particular, with firm productivity, thus undermining the causal interpretation of the productivity differential estimated above.

Following a standard approach, instrumental variable estimators were used to cope with this endogeneity issue. In line with the previous literature (Ciccone and Hall, 1996; Combes et al., 2008), the basic intuition is that history and geography may provide a source of exogenous spatial variation that affects the likelihood of having a city or an industrial district in a specific location. At the same time we expect these factors will be uncorrelated with current firm productivity in the manufacturing sector. Taking into account the discrete nature of the endogenous regressors, instruments for the UA and ID dummies were obtained taking the predicted value from a multinomial logit regression of LLMA type on a set of strictly exogenous or predetermined variables. Angrist and Pischke (2008, Sec. 4.6.1) show how this procedure can improve the fit of the instruments in the first stage, thus enhancing the precision of IV estimators.

The set of instrumental variables used in the first stage multinomial logit step includes the log of population density in 1921 and the share of population with an university or secondary school degree in 1971 (history), plus the share of LLMA's land near the coastline and the log of the LLMA average altitude (geography).

IV estimates, displayed in the third column of Table a1, not only confirm previous results but point to larger agglomeration effects on manufacturing productivity levels for both industrial districts and urban areas. To evaluate the dynamic pattern of productivity over the time span considered (1995-2006), the sample was split into two sub-periods. In line with evidence from the baseline model specification, it turns out that comparative advantages for urban areas remain stable while those of industrial districts show a tendency to decline over time (see Table a2 for detailed estimation results).

To single out aggregate TFP variation across differing LLMA types, in a final stage the other panel data dimensions were collapsed, yielding a single spatial cross-section featuring average TFP figures at the LLMA level. The aggregate TFP levels as defined in (3*a*) were first netted out of sectoral, size and statistical imputation effects by running the following regression:

$$\phi_{rst} = \alpha shflagimp_{rst} + \beta avfirmsize_{rst} + \lambda_s + \varepsilon_{rst} \quad (4a)$$

where *shflagimpg* and *avfirmsize* denote respectively the share of firms with imputed employment data and the average firm size in each stratum. Weighted least squares estimators were used to take account of the differences in the size of the strata.

Estimated residuals $\hat{\varepsilon}_{rst}$, obtained by fitting equation (4*a*) to the sample data, were subsequently averaged over industries using relative frequencies as weights, and these figures were finally averaged across years, yielding the desired aggregate TFP indicator at LLMA level, $\bar{\varepsilon}_r$. The latter was subsequently regressed on the ID and UA dummies plus geographical controls.

OLS and IV estimation results are displayed in Table a3. The TFP advantage of urban areas and industrial districts stand out even more sharply, especially in the case of IV estimates, which show the highest values across the different model specification considered here (a TFP gain of about 10 and 30 per cent for industrial districts and urban areas respectively).

The specifications outlined above were also estimated considering the sub-sample of small firms (those with below sector-year median employment level.). A twofold purpose motivates the exercise. First, we are interested in evaluating the case of small firms, as the theoretical literature has emphasized that in agglomerated areas they may benefit from external scale economies while remaining small. Second, our results on cities could be distorted by the presence of multi-plant firms. Usually these firms locate their corporate headquarters in big cities and their factories outside of urban areas. In our data set the local productivity advantages of these factories accrue to the urban area where the corporate headquarters are based, thereby distorting the assessment of a productivity premium in urban areas. To address this problem, we replicate the analysis by restricting the sample to firms with below sector-year median employment level, on the grounds that small firms usually are more likely to own a single plant.

Estimation results are reported in Tables a4 and a5 for the various specifications considered. Overall, the productivity advantage of urban areas and industrial districts is confirmed for the sub-sample of small firms, as is the ranking of urban areas and industrial districts.

On the whole, the robustness analysis carried out in this section confirms the ranking of the productivity advantages across areas as well as its evolution over time.

Table a1

Weighted Least Squares estimation of TFP at LLMA and sector level (standard errors in brackets) (1)					
ID	0.044***	0.023***	0.063 ***		
	(0.004)	(0.004)	(0.007)		
UA	0.180***	0.163***	0.250 ***		
	(0.008)	(0.007)	(0.014)		
Lsize	0.019***	0.027 ***	0.048 ***		
	(0.004)	(0.004)	(0.003)		
North-East	-0.004		-0.004		
	(0.005)		(0.004)		
Centre	-0.044 ***		-0.060 ***		
	(0.007)		(0.005)		
South	-0.274 ***		-0.275***		
	(0.007)		(0.006)		
Number of observations	46,094	46,094	46,094		
Adjusted R ²	0.884	0.886	0.792		

Sources: Based on Centrale dei Bilanci and Cerved data. (1) Standard errors are corrected for clustering at the level of individual LLMAs.

Table a2

Weighted Least Squares estimation of TFP at LLMA and sector level, by period					
(1995-2000 (with area dummies)	2001-2006 (with area dummies)	1995-2000 (with regional dummies)	2001-2006 (with regional dummies)	
ID	0.047***	0.040***	0.024***	0.023***	
	(0.005)	(0.005)	(0.006)	(0.006)	
UA	0.175***	0.184***	0.159***	0.168***	
	(0.010)	(0.001)	(0.010)	(0.011)	
Lsize	0.010	0.027***	0.020***	0.032***	
	(0.006)	(0.005)	(0.005)	(0.005)	
North-East	-0.005	-0.002			
	(0.006)	(0.007)			
Centre	-0.041***	-0.047***			
	(0.010)	(0.010)			
South	-0.293***	-0.259***			
	(0.010)	(0.009)			
Number of observations	22,275	23,819	22,275	23,819	
Adjusted R ²	0.892	0.877	0.895	0.879	

Sources: Based on Centrale dei Bilanci and Cerved data.

(1) Standard errors are corrected for clustering at the level of individual LLMAs.

Table a3

Estimation of TFP at LLMA level (standard errors in brackets) (1)				
	Weighted Least Squares	Instrumental variables		
ID	0.058**	0.114**		
	(0.021)	(0.042)		
UA	0.184**	0.332**		
	(0.067)	(0.111)		
North-East	-0.019	-0.020		
	(0.027)	(0.027)		
Centre	-0.050	-0.050		
	(0.028)	(0.028)		
South	-0.281 ***	-0.259 ***		
	(0.025)	(0.029)		
Number of observations	689	689		
Adjusted R ²	0.278	0.266		

Sources: Based on Centrale dei Bilanci and Cerved data. (1) Standard errors are corrected for clustering at the level of individual LLMAs.

Weighted Least Squares estimation of TFP at LLMA and sector level; small firm sample (1) (standard errors in brackets) (2)

	With area dummies	With regional dummies	Instrumental variables
ID	0.028 ***	0.018***	0.055***
	(0.003)	(0.003)	(0.008)
UA	0.109***	0.099***	0.173***
	(0.004)	(0.004)	(0.014)
Lsize	0.040 ***	0.037 ***	0.048***
	(0.006)	(0.006)	(0.006)
North-East	-0.017***		
	(0.003)		
Centre	-0.051 ***		
	(0.004)		
South	-0.258 ***		
	(0.005)		
Number of observations	35,755	35,755	35,755
Adjusted R ²	0.885	0.866	0.773

Sources: Based on Centrale dei Bilanci and Cerved data. (1) Small firms are those with below sector-year median employment level. - (2) Standard errors are corrected for clustering at the level of individual LLMAs.

Weighted Least Squares estimation of TFP at LLMA and sector level; small firm sample, by period (1) (standard errors in brackets) (2)

	1995-2000 (with area dummies)	2001-2006 (with area dummies)	1995-2000 (with regional dummies)	2001-2006 (with regional dummies)
ID	0.037***	0.020***	0.024***	0.013***
	(0.004)	(0.004)	(0.005)	(0.005)
UA	0.108***	0.110***	0.099***	0.101***
	(0.005)	(0.005)	(0.005)	(0.005)
Lsize	0.035***	0.043***	0.032***	0.042***
	(0.009)	(0.009)	(0.009)	(0.009)
North-East	-0.013***	-0.020***		
	(0.004)	(0.004)		
Centre	-0.045***	-0.056***		
	(0.006)	(0.005)		
South	-0.283***	-0.236***		
	(0.007)	(0.007)		
Number of observations	17,295	18,460	17,295	18,460
Adjusted R ²	0.889	0.882	0.891	0.883

Sources: Based on Centrale dei Bilanci and Cerved data. (1) Small firms are those with below sector-year median employment level. - (2) Standard errors are corrected for clustering at the level of individual LLMAs.

References

- Accetturo, A., V. Di Giacinto, G. Micucci and M. Pagnini (2011), "Local Productivity Differences Through Thick and Thin: Market Size, Entry Costs and Openness to Trade", mimeo.
- Angrist, J.D. and J. Pischke (2008), *Mostly Harmless Econometrics: An Empiricist's Companion*, Princeton University Press.
- Arimoto, Y., K. Nakajima and T. Okazaki (2009), "Agglomeration or Selection? The Case of the Japanese Silk-reeling Industry, 1909-1916", RIETI Discussion Paper Series 10-E-003.
- Baldwin, R., Okubo, T., (2006), "Heterogeneous firms, agglomeration and economic geography: spatial selection and sorting", *Journal of Economic Geography*, 6, 323–346.
- Becattini, G. (1990), "The Marshallian industrial district as a socio-economic notion", in F. Pyke, G. Becattini and W. Sengenberger (eds.), *Industrial Districts and Inter-firm Co-operation in Italy*, Geneva: International Labor Office, pp. 37-51.
- Becchetti, L. and A. Castelli (2005), "Inside the black box: economic performance and technology adoption when space and product relationships matter", *Rivista di Politica Economica*, 95 (1-2), 137-175.
- Becchetti, L. and S. Rossi (2000), "The positive effect of industrial district on the export performance of Italian firms", *Review of Industrial Organization*, 16 (1), 53-68.
- Bond, S., J. Elston, J. Mairesse and B. Mulkay (1997), "Financial Factors and Investment in Belgium, France, Germany and the UK: A Comparison Using Company Panel Data", NBER Working Papers, no. 5900.
- Brandolini, A. and M. Bugamelli (eds.) (2009), "Rapporto sulle tendenze del sistema produttivo italiano", Bank of Italy Occasional Papers, no. 45, Rome.
- Briant, A. (2010), "Marshall's scale economies: a quantile regression approach", mimeo.
- Bronzini, R. (2000), "Sistemi produttivi locali e commercio estero: un'analisi territoriale delle esportazioni italiane", in L.F. Signorini (ed.), Lo sviluppo locale. Un'indagine della Banca d'Italia sui distretti industriali, Corigliano Calabro: Meridiana Libri.
- Bugamelli, M. and L. Infante (2005), "Sunk costs of exports: a role for industrial districts?", in Banca d'Italia, *Local Economies and Internationalization in Italy*, Rome, Banca d'Italia, pp. 343-372.
- Cainelli, G. and N. De Liso (2005), "Innovation in Industrial Districts: Evidence from Italy", *Industry and Innovation*, vol. 12, n. 3, pp. 383-398.
- Cannari, L., M. Magnani and G. Pellegrini (2009), "Quali politiche per il Sud? Il ruolo delle politiche nazionali e regionali nell'ultimo decennio", Bank of Italy Occasional Papers, no. 50 Rome.
- Carabelli, A., R. Rabellotti and G. Hirsch (2009), "Italian Industrial Districts on the Move: Where Are They Going?", *European Planning Studies*, 17:1, pp. 19-41.
- Castellani, D. and G. Giovannetti (2010), "Productivity and the international firm: dissecting heterogeneity", *Journal of Economic Policy Reform*, vol. 13, n. 1, March, pp. 25–42.

- Ciccone, A. and R.E. Hall (1996), "Productivity and the Density of Economic Activity", *American Economic Review*, vol. 86(1), pp. 54-70.
- Cingano, F. and F. Schivardi (2005), "Identifying the sources of local productivity growth", in Banca d'Italia, *Local Economies and Internationalization in Italy*, Rome.
- Combes P-P., Duranton G. and L. Gobillon (2008). "Spatial wage disparities: Sorting matters!," Journal of Urban Economics, vol. 63(2), p. 723-742
- Combes, P., G. Duranton, L. Gobillon and S. Roux (2008), "Estimating Agglomeration Economies with History, Geology, and Worker Effects", CEPR Discussion Papers 6728.
- Combes, P., G. Duranton, L. Gobillon, D. Puga and S. Roux (2009), "The productivity advantages of large cities: Distinguishing agglomeration from firm selection", CEPR Discussion Paper no. 7191.
- Del Gatto, M., G. Ottaviano and M. Pagnini (2008), "Openness to trade and industry cost dispersion: Evidence from a panel of Italian firms", *Journal of Regional Science*, Vol. 48, No. 1, pp. 97–129.
- Di Addario, S. and E. Patacchini (2008) "Wages and the City. Evidence from Italy", Labor Economics, 15, 1040-1061.
- Duranton, G. (2011), "California Dreamin': the Feeble Case for Cluster Policies", Review of *Economic Analysis*, no. 3, pp. 3-45.
- Duranton, G., and D. Puga (2001), "Nursery cities: Urban diversity, process innovation, and the life cycle of products", *American Economic Review* 91 (5), 1454–1477.
- Fabiani, S., G. Pellegrini, E. Romagnano and L.F. Signorini (2000), "Efficiency and localisation: the case of Italian districts", in M. Bagella and L. Becchetti (eds.), *The Competitive Advantage of Industrial Districts: Theoretical and Empirical Analysis*, Heidelberg: Phisica-Verlag.
- Foresti, G., F. Guelpa and S. Trenti (2009), "Effetto distretto: esiste ancora?", Intesa Sanpaolo, Collana Ricerche, no. R09-01.
- Glaeser, E.L, S.S. Rosenthal and W.C. Strange (2009), "Urban Economics and Entrepreneurship", NBER Working Paper No. 15536.
- Glaeser, E.L. and G.A.M. Ponzetto (2010), "Did the Death of Distance Hurt Detroit and Help New York?", in EL. Glaeser (ed.), *Agglomeration Economies*, National Bureau of Economic Research, The University of Chicago Press.
- Gola, C. and A. Mori (2000), "Concentrazione spaziale della produzione e specializzazione internazionale dell'industria italiana", in L.F. Signorini (ed.), Lo sviluppo locale. Un'indagine della Banca d'Italia sui distretti industriali, Corigliano Calabro: Meridiana Libri.
- Holmes, T.J. and J.J. Stevens. 2010. "An Alternative Theory of the Plant Size Distribution with an Application to Trade," typescript, University of Minnesota.
- Iuzzolino, G. and G. Micucci (2011), "Le recenti trasformazioni dei distretti industriali italiani", in *Secondo Rapporto Nazionale dell'Osservatorio sui Distretti*, Federazione dei Distretti Italiani, Venezia.
- Koenker, R. and J.G. Bassett (1978) "Regression quantiles", Econometrica 46, 33-50.
- Levinsohn, J. and A. Petrin (2003), "Estimating production functions using inputs to control for unobservables", *Review of Economic Studies* 70(2): 317–342.

- Matano A. and P. Naticchioni (2011), "Wage Distribution and the Spatial Sorting of Workers". Journal of Economic Geography, doi: 10.1093/jeg/Ibr013
- Melitz, M. (2003). "The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity", *Econometrica*, 71(6): 1695-1725.
- Melitz, M. and G. Ottaviano (2008) "Market size, trade, and productivity", Review of Economic Studies, 75: 295–316.
- Melo, P., D. Graham and R. Noland (2009), "A meta-analysis of estimates of urban agglomeration economies," Regional Science and Urban Economics, 39, 332-342.
- Nocke, V. (2006), "A gap for me: Entrepreneurs and entry", Journal of the European Economic Association 4, 929-956.
- Okubo, T., M.P. Picard and J. F. Thisse (2010), "The spatial selection of heterogeneous firms", *Journal of International Economics*, 82, 230–237.
- Olley, G.S. and A. Pakes (1996), "The Dynamics of Productivity in the Telecommunications Equipment Industry." *Econometrica*, 64(6), 1263-97.
- Rosenthal, S.S., and W.C. Strange (2004), "Evidence on the nature and sources of agglomeration economies", in V. Henderson and J.-F. Thisse (eds.), *Handbook of Regional and Urban Economics*, vol. 4, Amsterdam: North-Holland.
- Signorini, L.F. (1994), "The price of Prato, or measuring the industrial district effect", *Papers in Regional Science*, 73 (4), 369-392.
- Syverson, C. (2004a), "Market Structure and Productivity: A Concrete Example", *Journal of Political Economy*, vol. 112, no. 6.
- Syverson, C. (2004b), "Product Substitutability and Productivity Dispersion", Review of *Economics and Statistics*, 86, 534–550.
- Syverson, C. (2010), "What determines Productivity?", NBER Working Paper No. 15712.
- Venables, A. J. (2011), "Productivity in cities: self-selection and sorting", Journal of Economic Geography, 11(2), 241-51.

RECENTLY PUBLISHED "TEMI" (*)

- N. 818 *Leaving home and housing prices. The experience of Italian youth emancipation*, by Francesca Modena and Concetta Rondinelli (September 2011).
- N. 819 The interbank market after the financial turmoil: squeezing liquidity in a "lemons market" or asking liquidity "on tap", by Antonio De Socio (September 2011).
- N. 820 The relationship between the PMI and the Italian index of industrial production and the impact of the latest economic crisis, by Valentina Aprigliano (September 2011).
- N. 821 Inside the sovereign credit default swap market: price discovery, announcements, market behaviour and corporate sector, by Alessandro Carboni (September 2011).
- N. 822 The demand for energy of Italian households, by Ivan Faiella (September 2011).
- N. 823 Sull'ampiezza ottimale delle giurisdizioni locali: il caso delle province italiane, by Guglielmo Barone (September 2011).
- N. 824 *The public-private pay gap: a robust quantile approach*, by Domenico Depalo and Raffaela Giordano (September 2011).
- N. 825 *Evaluating students' evaluations of professors*, by Michele Braga, Marco Paccagnella and Michele Pellizzari (October 2011).
- N. 826 Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy, by Massimiliano Affinito (October 2011).
- N. 827 Foreign trade, home linkages and the spatial transmission of economic fluctuations in Italy, by Valter Di Giacinto (October 2011).
- N. 828 *Healthcare in Italy: expenditure determinants and regional differentials*, by Maura Francese and Marzia Romanelli (October 2011).
- N. 829 Bank heterogeneity and interest rate setting: what lessons have we learned since Lehman Brothers?, by Leonardo Gambacorta and Paolo Emilio Mistrulli (October 2011).
- N. 830 Structural reforms and macroeconomic performance in the euro area countries: a model-based assessment, by Sandra Gomes, Pascal Jacquinot, Matthias Mohr and Massimiliano Pisani (October 2011).
- N. 831 *Risk measures for autocorrelated hedge fund returns*, by Antonio Di Cesare, Philip A. Stork and Casper G. de Vries (October 2011).
- N. 832 *Investment forecasting with business survey data*, by Leandro D'Aurizio and Stefano Iezzi (November 2011).
- N. 833 *Electoral rules and voter turnout*, by Guglielmo Barone and Guido de Blasio (November 2011).
- N. 834 A method to estimate power parameter in Exponential Power Distribution via polynomial regression, by Daniele Coin (November 2011).
- N. 835 Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices in the US, by Olivier Blanchard and Marianna Riggi (November 2011).
- N. 836 *Bayesian analysis of coefficient instability in dynamic regressions*, by Emanuela Ciapanna and Marco Taboga (November 2011).
- N. 837 The effects of financial and real wealth on consumption: new evidence from OECD countries, by Riccardo De Bonis and Andrea Silvestrini (November 2011).
- N. 838 *Households' savings in China*, by Riccardo Cristadoro and Daniela Marconi (November 2011).
- N. 839 The effects of fiscal shocks with debt-stabilizing budgetary policies in Italy, by Francesco Caprioli and Sandro Momigliano (November 2011).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2009

- F. PANETTA, F. SCHIVARDI and M. SHUM, *Do mergers improve information? Evidence from the loan market*, Journal of Money, Credit, and Banking, v. 41, 4, pp. 673-709, **TD No. 521 (October 2004).**
- M. BUGAMELLI and F. PATERNÒ, *Do workers' remittances reduce the probability of current account reversals?*, World Development, v. 37, 12, pp. 1821-1838, **TD No. 573 (January 2006).**
- P. PAGANO and M. PISANI, *Risk-adjusted forecasts of oil prices*, The B.E. Journal of Macroeconomics, v. 9, 1, Article 24, **TD No. 585 (March 2006).**
- M. PERICOLI and M. SBRACIA, The CAPM and the risk appetite index: theoretical differences, empirical similarities, and implementation problems, International Finance, v. 12, 2, pp. 123-150, TD No. 586 (March 2006).
- R. BRONZINI and P. PISELLI, *Determinants of long-run regional productivity with geographical spillovers: the role of R&D, human capital and public infrastructure,* Regional Science and Urban Economics, v. 39, 2, pp.187-199, **TD No. 597 (September 2006).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and the business cycle*, Journal of Financial Stability, v. 5, 4, pp. 393-409, **TD No. 601 (September 2006).**
- F. BALASSONE, D. FRANCO and S. ZOTTERI, *The reliability of EMU fiscal indicators: risks and safeguards*, in M. Larch and J. Nogueira Martins (eds.), Fiscal Policy Making in the European Union: an Assessment of Current Practice and Challenges, London, Routledge, **TD No. 633 (June 2007).**
- A. CIARLONE, P. PISELLI and G. TREBESCHI, *Emerging Markets' Spreads and Global Financial Conditions*, Journal of International Financial Markets, Institutions & Money, v. 19, 2, pp. 222-239, **TD No. 637 (June 2007)**.
- S. MAGRI, *The financing of small innovative firms: the Italian case*, Economics of Innovation and New Technology, v. 18, 2, pp. 181-204, **TD No. 640 (September 2007).**
- V. DI GIACINTO and G. MICUCCI, The producer service sector in Italy: long-term growth and its local determinants, Spatial Economic Analysis, Vol. 4, No. 4, pp. 391-425, TD No. 643 (September 2007).
- F. LORENZO, L. MONTEFORTE and L. SESSA, *The general equilibrium effects of fiscal policy: estimates for the euro area*, Journal of Public Economics, v. 93, 3-4, pp. 559-585, **TD No. 652** (November 2007).
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS, *Securitisation and the bank lending channel*, European Economic Review, v. 53, 8, pp. 996-1009, **TD No. 653** (November 2007).
- R. GOLINELLI and S. MOMIGLIANO, *The Cyclical Reaction of Fiscal Policies in the Euro Area. A Critical Survey of Empirical Research*, Fiscal Studies, v. 30, 1, pp. 39-72, **TD No. 654 (January 2008).**
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "Inflation Perceptions"? A survey-based analysis of Italian consumers, Giornale degli Economisti e Annali di Economia, v. 68, 1, pp. 25-52, TD No. 655 (January 2008).
- F. MACCHERONI, M. MARINACCI, A. RUSTICHINI and M. TABOGA, *Portfolio selection with monotone mean*variance preferences, Mathematical Finance, v. 19, 3, pp. 487-521, **TD No. 664 (April 2008).**
- M. AFFINITO and M. PIAZZA, What are borders made of? An analysis of barriers to European banking integration, in P. Alessandrini, M. Fratianni and A. Zazzaro (eds.): The Changing Geography of Banking and Finance, Dordrecht Heidelberg London New York, Springer, TD No. 666 (April 2008).
- A. BRANDOLINI, On applying synthetic indices of multidimensional well-being: health and income inequalities in France, Germany, Italy, and the United Kingdom, in R. Gotoh and P. Dumouchel (eds.), Against Injustice. The New Economics of Amartya Sen, Cambridge, Cambridge University Press, TD No. 668 (April 2008).
- G. FERRERO and A. NOBILI, *Futures contract rates as monetary policy forecasts*, International Journal of Central Banking, v. 5, 2, pp. 109-145, **TD No. 681 (June 2008).**
- P. CASADIO, M. LO CONTE and A. NERI, Balancing work and family in Italy: the new mothers' employment decisions around childbearing, in T. Addabbo and G. Solinas (eds.), Non-Standard Employment and Qualità of Work, Physica-Verlag. A Sprinter Company, TD No. 684 (August 2008).
- L. ARCIERO, C. BIANCOTTI, L. D'AURIZIO and C. IMPENNA, *Exploring agent-based methods for the analysis* of payment systems: A crisis model for StarLogo TNG, Journal of Artificial Societies and Social Simulation, v. 12, 1, **TD No. 686 (August 2008).**
- A. CALZA and A. ZAGHINI, Nonlinearities in the dynamics of the euro area demand for M1, Macroeconomic Dynamics, v. 13, 1, pp. 1-19, **TD No. 690 (September 2008).**
- L. FRANCESCO and A. SECCHI, *Technological change and the households' demand for currency*, Journal of Monetary Economics, v. 56, 2, pp. 222-230, **TD No. 697 (December 2008).**

- G. ASCARI and T. ROPELE, *Trend inflation, taylor principle, and indeterminacy*, Journal of Money, Credit and Banking, v. 41, 8, pp. 1557-1584, **TD No. 708** (May 2007).
- S. COLAROSSI and A. ZAGHINI, *Gradualism, transparency and the improved operational framework: a look at overnight volatility transmission*, International Finance, v. 12, 2, pp. 151-170, **TD No. 710 (May 2009).**
- M. BUGAMELLI, F. SCHIVARDI and R. ZIZZA, *The euro and firm restructuring*, in A. Alesina e F. Giavazzi (eds): Europe and the Euro, Chicago, University of Chicago Press, **TD No. 716 (June 2009).**
- B. HALL, F. LOTTI and J. MAIRESSE, *Innovation and productivity in SMEs: empirical evidence for Italy*, Small Business Economics, v. 33, 1, pp. 13-33, **TD No. 718 (June 2009).**

2010

- A. PRATI and M. SBRACIA, *Uncertainty and currency crises: evidence from survey data*, Journal of Monetary Economics, v, 57, 6, pp. 668-681, **TD No. 446 (July 2002).**
- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, v. 42, 19-21, pp. 2399-2415, **TD No. 458 (December 2002).**
- S. MAGRI, *Debt maturity choice of nonpublic Italian firms*, Journal of Money, Credit, and Banking, v.42, 2-3, pp. 443-463, **TD No. 574 (January 2006).**
- G. DE BLASIO and G. NUZZO, *Historical traditions of civicness and local economic development*, Journal of Regional Science, v. 50, 4, pp. 833-857, **TD No. 591 (May 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, v. 62, 2, pp. 374-394, **TD No. 598 (September 2006).**
- S. NERI and A. NOBILI, *The transmission of US monetary policy to the euro area,* International Finance, v. 13, 1, pp. 55-78, **TD No. 606 (December 2006).**
- F. ALTISSIMO, R. CRISTADORO, M. FORNI, M. LIPPI and G. VERONESE, *New Eurocoin: Tracking Economic Growth in Real Time*, Review of Economics and Statistics, v. 92, 4, pp. 1024-1034, **TD No. 631 (June 2007).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and taxation*, Journal of Banking and Finance, v. 34, 11, pp. 2801-2810, **TD No. 649** (November 2007).
- M. IACOVIELLO and S. NERI, *Housing market spillovers: evidence from an estimated DSGE model,* American Economic Journal: Macroeconomics, v. 2, 2, pp. 125-164, **TD No. 659 (January 2008).**
- F. BALASSONE, F. MAURA and S. ZOTTERI, *Cyclical asymmetry in fiscal variables in the EU*, Empirica, **TD** No. 671, v. 37, 4, pp. 381-402 (June 2008).
- F. D'AMURI, O. GIANMARCO I.P. and P. GIOVANNI, The labor market impact of immigration on the western german labor market in the 1990s, European Economic Review, v. 54, 4, pp. 550-570, TD No. 687 (August 2008).
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, v. 89, 1, pp. 173-190, **TD No. 688 (September 2008).**
- S. NOBILI and G. PALAZZO, *Explaining and forecasting bond risk premiums*, Financial Analysts Journal, v. 66, 4, pp. 67-82, **TD No. 689 (September 2008).**
- A. B. ATKINSON and A. BRANDOLINI, *On analysing the world distribution of income*, World Bank Economic Review, v. 24, 1, pp. 1-37, **TD No. 701 (January 2009).**
- R. CAPPARIELLO and R. ZIZZA, Dropping the Books and Working Off the Books, Labour, v. 24, 2, pp. 139-162, **TD No. 702 (January 2009).**
- C. NICOLETTI and C. RONDINELLI, *The (mis)specification of discrete duration models with unobserved heterogeneity: a Monte Carlo study*, Journal of Econometrics, v. 159, 1, pp. 1-13, **TD No. 705** (March 2009).
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector: the case of Italy*, Macroeconomic Dynamics, v. 14, 5, pp. 677-708, **TD No. 706** (March 2009).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Dynamic macroeconomic effects of public capital: evidence from regional Italian data, Giornale degli economisti e annali di economia, v. 69, 1, pp. 29-66, TD No. 733 (November 2009).
- F. COLUMBA, L. GAMBACORTA and P. E. MISTRULLI, *Mutual Guarantee institutions and small business finance*, Journal of Financial Stability, v. 6, 1, pp. 45-54, **TD No. 735** (November 2009).
- A. GERALI, S. NERI, L. SESSA and F. M. SIGNORETTI, *Credit and banking in a DSGE model of the Euro Area,* Journal of Money, Credit and Banking, v. 42, 6, pp. 107-141, **TD No. 740 (January 2010).**
- M. AFFINITO and E. TAGLIAFERRI, Why do (or did?) banks securitize their loans? Evidence from Italy, Journal

of Financial Stability, v. 6, 4, pp. 189-202, TD No. 741 (January 2010).

- S. FEDERICO, Outsourcing versus integration at home or abroad and firm heterogeneity, Empirica, v. 37, 1, pp. 47-63, **TD No. 742** (February 2010).
- V. DI GIACINTO, *On vector autoregressive modeling in space and time*, Journal of Geographical Systems, v. 12, 2, pp. 125-154, **TD No. 746 (February 2010).**
- L. FORNI, A. GERALI and M. PISANI, *The macroeconomics of fiscal consolidations in euro area countries,* Journal of Economic Dynamics and Control, v. 34, 9, pp. 1791-1812, **TD No. 747** (March 2010).
- S. MOCETTI and C. PORELLO, *How does immigration affect native internal mobility? new evidence from Italy*, Regional Science and Urban Economics, v. 40, 6, pp. 427-439, **TD No. 748 (March 2010)**.
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap spread changes before and during the subprime financial turmoil, Journal of Current Issues in Finance, Business and Economics, v. 3, 4, pp., **TD No. 749** (March 2010).
- P. CIPOLLONE, P. MONTANARO and P. SESTITO, Value-added measures in Italian high schools: problems and findings, Giornale degli economisti e annali di economia, v. 69, 2, pp. 81-114, TD No. 754 (March 2010).
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, *Asset-based measurement of poverty*, Journal of Policy Analysis and Management, v. 29, 2, pp. 267-284, **TD No. 755** (March 2010).
- G. CAPPELLETTI, A Note on rationalizability and restrictions on beliefs, The B.E. Journal of Theoretical Economics, v. 10, 1, pp. 1-11, **TD No. 757** (April 2010).
- S. DI ADDARIO and D. VURI, Entrepreneurship and market size. the case of young college graduates in Italy, Labour Economics, v. 17, 5, pp. 848-858, **TD No. 775 (September 2010).**
- A. CALZA and A. ZAGHINI, *Sectoral money demand and the great disinflation in the US*, Journal of Money, Credit, and Banking, v. 42, 8, pp. 1663-1678, **TD No. 785 (January 2011).**

2011

- S. DI ADDARIO, *Job search in thick markets*, Journal of Urban Economics, v. 69, 3, pp. 303-318, **TD No.** 605 (December 2006).
- F. SCHIVARDI and E. VIVIANO, *Entry barriers in retail trade*, Economic Journal, v. 121, 551, pp. 145-170, **TD** No. 616 (February 2007).
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing excess liquidity in the Euro Area: the role of sectoral distribution of money, Applied Economics, v. 43, 23, pp. 3213-3230, **TD No. 627** (April 2007).
- P. E. MISTRULLI, Assessing financial contagion in the interbank market: maximum entropy versus observed interbank lending patterns, Journal of Banking & Finance, v. 35, 5, pp. 1114-1127, TD No. 641 (September 2007).
- E. CIAPANNA, Directed matching with endogenous markov probability: clients or competitors?, The RAND Journal of Economics, v. 42, 1, pp. 92-120, **TD No. 665 (April 2008).**
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, v. 78, 311, pp. 480-500, **TD No. 673 (June 2008).**
- V. DI GIACINTO e M. PAGNINI, Local and global agglomeration patterns: two econometrics-based indicators, Regional Science and Urban Economics, v. 41, 3, pp. 266-280, **TD No. 674 (June 2008)**.
- G. BARONE and F. CINGANO, Service regulation and growth: evidence from OECD countries, Economic Journal, v. 121, 555, pp. 931-957, TD No. 675 (June 2008).
- R. GIORDANO and P. TOMMASINO, What determines debt intolerance? The role of political and monetary *institutions*, European Journal of Political Economy, v. 27, 3, pp. 471-484, **TD No. 700 (January 2009).**
- P. ANGELINI, A. NOBILI e C. PICILLO, *The interbank market after August 2007: What has changed, and why?*, Journal of Money, Credit and Banking, v. 43, 5, pp. 923-958, **TD No. 731 (October 2009).**
- L. FORNI, A. GERALI and M. PISANI, *The Macroeconomics of Fiscal Consolidation in a Monetary Union: the Case of Italy*, in Luigi Paganetto (ed.), Recovery after the crisis. Perspectives and policies, VDM Verlag Dr. Muller, **TD No. 747 (March 2010).**
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap changes before and during the subprime financial turmoil, in Barbara L. Campos and Janet P. Wilkins (eds.), The Financial Crisis: Issues in Business, Finance and Global Economics, New York, Nova Science Publishers, Inc., **TD No. 749 (March 2010).**
- A. LEVY and A. ZAGHINI, *The pricing of government guaranteed bank bonds*, Banks and Bank Systems, v. 6, 3, pp. 16-24, **TD No. 753 (March 2010).**

- G. GRANDE and I. VISCO, A public guarantee of a minimum return to defined contribution pension scheme members, The Journal of Risk, v. 13, 3, pp. 3-43, **TD No. 762 (June 2010).**
- P. DEL GIOVANE, G. ERAMO and A. NOBILI, *Disentangling demand and supply in credit developments: a survey-based analysis for Italy*, Journal of Banking and Finance, v. 35, 10, pp. 2719-2732, **TD No.** 764 (June 2010).
- G. BARONE and S. MOCETTI, With a little help from abroad: the effect of low-skilled immigration on the female labour supply, Labour Economics, v. 18, 5, pp. 664-675, **TD No. 766 (July 2010).**
- A. FELETTIGH and S. FEDERICO, *Measuring the price elasticity of import demand in the destination markets of italian exports*, Economia e Politica Industriale, v. 38, 1, pp. 127-162, **TD No. 776 (October 2010).**
- S. MAGRI and R. PICO, *The rise of risk-based pricing of mortgage interest rates in Italy*, Journal of Banking and Finance, v. 35, 5, pp. 1277-1290, **TD No. 778 (October 2010).**
- M. TABOGA, Under/over-valuation of the stock market and cyclically adjusted earnings, International Finance, v. 14, 1, pp. 135-164, **TD No. 780 (December 2010).**
- S. NERI, *Housing, consumption and monetary policy: how different are the U.S. and the Euro area?*, Journal of Banking and Finance, v.35, 11, pp. 3019-3041, **TD No. 807** (April 2011).
- V. CUCINIELLO, *The welfare effect of foreign monetary conservatism with non-atomistic wage setters*, Journal of Money, Credit and Banking, v. 43, 8, pp. 1719-1734, **TD No. 810 (June 2011).**
- A. CALZA and A. ZAGHINI, welfare costs of inflation and the circulation of US currency abroad, The B.E. Journal of Macroeconomics, v. 11, 1, Art. 12, **TD No. 812 (June 2011).**
- I. FAIELLA, *La spesa energetica delle famiglie italiane*, Energia, v. 32, 4, pp. 40-46, **TD No. 822 (September 2011).**
- R. DE BONIS and A. SILVESTRINI, *The effects of financial and real wealth on consumption: new evidence from* OECD countries, Applied Financial Economics, v. 21, 5, pp. 409–425, **TD No. 837 (November 2011).**

2012

A. ACCETTURO and G. DE BLASIO, Policies for local development: an evaluation of Italy's "Patti Territoriali", Regional Science and Urban Economics, v. 42, 1-2, pp. 15-26, TD No. 789 (January 2006).

FORTHCOMING

- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD No.** 578 (February 2006).
- F. CINGANO and A. ROSOLIA, *People I know: job search and social networks*, Journal of Labor Economics, **TD No. 600 (September 2006).**
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, **TD No. 691 (September 2008).**
- P. SESTITO and E. VIVIANO, *Reservation wages: explaining some puzzling regional patterns*, Labour, **TD No. 696 (December 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, **TD No. 698 (December 2008).**
- F. LIPPI and A. NOBILI, *Oil and the macroeconomy: a quantitative structural analysis*, Journal of European Economic Association, **TD No. 704** (March 2009).
- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, **TD No. 709 (May 2009).**
- Y. ALTUNBAS, L. GAMBACORTA, and D. MARQUÉS-IBÁÑEZ, *Bank risk and monetary policy*, Journal of Financial Stability, **TD No. 712 (May 2009).**
- G. BARONE and S. MOCETTI, *Tax morale and public spending inefficiency*, International Tax and Public Finance, **TD No. 732 (November 2009).**
- I. BUONO and G. LALANNE, *The effect of the Uruguay Round on the intensive and extensive margins of trade*, Journal of International Economics, **TD No. 835 (February 2011).**
- G. BARONE, R. FELICI and M. PAGNINI, *Switching costs in local credit markets*, International Journal of Industrial Organization, **TD No. 760 (June 2010).**

- E. COCOZZA and P. PISELLI, Testing for east-west contagion in the European banking sector during the financial crisis, in R. Matoušek; D. Stavárek (eds.), Financial Integration in the European Union, Taylor & Francis, TD No. 790 (February 2011).
- S. NERI and T. ROPELE, *Imperfect information, real-time data and monetary policy in the Euro area,* The Economic Journal, **TD No. 802 (March 2011).**
- M. AFFINITO, Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy, Journal of Banking and Finance, **TD No. 826 (October 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, **TD No. 835 (November 2011).**
- R. CRISTADORO and D. MARCONI, *Households Savings in China*, Chinese Economic and Business Studies, **TD No. 838 (November 2011).**